

## CER-ETH - Center of Economic Research at ETH Zurich

## Behavioral Anomalies and Fuel Efficiency: Evidence from Motorcycles in Nepal

M. Filippini, N. Kumar, S. Srinivasan

Working Paper 21/353 May 2021

**Economics Working Paper Series** 



Eidgenössische Technische Hochschule Zürich Swiss Federal Institute of Technology Zurich

# Behavioral Anomalies and Fuel Efficiency: Evidence from Motorcycles in Nepal

Massimo Filippini<sup>1,2</sup>, Nilkanth Kumar<sup>1</sup>, and Suchita Srinivasan\*<sup>1</sup>

<sup>1</sup>Center of Economic Research (CER-ETH), ETH Zürich, Switzerland <sup>2</sup>Università della Svizzera italiana, Switzerland

Last revision: May 19, 2021

#### **Abstract**

Air pollution is a grave problem in urban areas of developing countries, with the transport sector being one of the largest contributors to emissions. A possibility to reduce carbon dioxide emissions would be for individuals to switch to more fuel-efficient vehicles. However, a gamut of behavioral anomalies and market failures have been known to inhibit individuals from investing in fuel-efficiency (due to the well-known 'energy-efficiency gap'). In this study, we use novel data from Kathmandu, Nepal to understand the socio-economic and psychological determinants of three behavioral anomalies, namely present bias, loss aversion, risk aversion, as well as time preferences. In a second step, we evaluate the effect of these anomalies on the energy-efficiency gap in the choice of motorcycles of individuals. We find that present-biased individuals are less likely to invest in fuel-efficient motorcycles, and thus more likely to buy motorcycles having relatively high total lifetime costs. We also find that other factors such as income, as well as having applied for loans, play an important role in determining these choices. Our results suggest that behavioral anomalies may indeed pose as a hindrance to individuals making cost-minimizing (and also environmentally sound) investment decisions.

JEL Classification: D1, D8; Q4; Q5

**Keywords:** Behavioral anomalies; Present bias; Fuel efficiency; Energy-efficiency gap; Motorcycles; Nepal

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

<sup>\*</sup>Corresponding author at Center of Economic Research (CER-ETH), ETH Zürich, Zürichbergstrasse 18, 8032 Zürich, Switzerland. Phone: +41 44 632 65 34, Fax: +41 44 632 10 50. <suchitas@ethz.ch>.

## Acknowledgements

This research is partly funded by an ETH R4D seed money grant. We are grateful to our local partner, Facts Nepal, for collaborating with us in conducting the survey. Facts Nepal was not responsible for the survey design, analysis and interpretation of data. We also thank Nirmal Kumar Raut at Tribhuvan University in Kathmandu, Nepal for his feedback, local outreach and support during the project, as well as participants during the FSR Annual Climate Conference for their useful feedback. All omissions and errors remain our responsibility.

#### 1 Introduction

Air pollution is the cause for almost seven million deaths globally every year according to the World Health Organization, out which 4.2 million deaths can be attributed to outdoor air pollution (World Health Organization, 2016). While the negative health implications of excessive levels of air pollution are widespread, urban areas in low and middle-income countries (LMICs) bear the brunt: according to statistics from the WHO Urban Air Quality Database, 98% of cities in LMICs with more than 100,000 inhabitants do not meet WHO air quality guidelines (this share is 56% in high-income countries) (World Health Organization, 2016). These alarming statistics, combined with the fact that these countries are often ill-equipped to deal with the economic costs of air pollution, necessitates identifying the sectors having high local and global negative externalities, as well as understanding the reasons.

Kathmandu, Nepal is an example of one such city, where air pollution poses a grave threat to public health: the Environmental Performance Index (EPI) of Nepal's air quality ranked 176th out of 180 countries (Wendling et al., 2018), while an individual living in Kathmandu can expect to gain up to 4.7 years of life if concentrations of  $PM_{2.5}$  are reduced to the WHO guidelines (Energy Policy Institute at the University of Chicago, 2019). One of the primary sources of air pollution in Kathmandu is the transport sector. Inefficient transport contributes to approximately 63% of all  $PM_{10}$  emissions in the Kathmandu valley, for example (Schwela, 2009).

Improving the efficiency, reliability as well as quality of public transport remains an important objective in many cities in developing countries. However, given the large share of two-wheelers in the vehicle fleet in Kathmandu (including motorcycles, internal combustion engine (ICE)-based scooters and electric scooters, which accounted for about 80% of vehicle sales in 2016-17 (Facts Research and Analytics, 2018)), improving the efficiency of private transport is pivotal in reducing greenhouse gas emissions, as well as air pollution, from private transport.

As of 2010, about 25% of two-wheelers in Kathmandu still used pre-Euro standard engine technologies (Shrestha et al., 2013). In 2010, they consumed the largest share of petrol (about 25%) compared to other vehicle types in the Kathmandu valley (Bajracharya and Bhattarai, 2016). Moreover, the two-wheeler fleet in Kathmandu was also the largest contributor to other greenhouse gas emissions, such as those of VOC, CO and CH<sub>4</sub> (Shrestha et al., 2013). These statistics suggest the possibility that two-wheelers operating in Kathmandu are likely to be inefficient. Hereafter, we will use the term 'motorcycle' in order to refer to both the traditional motorcycles, as well as ICE-based scooters (but not electric scooters, which are not the focus of this study).

For instance, we find that there is significant variation in the fuel efficiency (measured in terms of kilometres per litre of petrol, or km/l) of motorcycles within a given engine size (measured in cubic centimetres, or 'cc') in Nepal, which is reflected in our data.<sup>1</sup> These differences in fuel economy can lead to substantial differences not just in emissions, but also in terms of annual fuel costs, depending on usage.<sup>2</sup> Moreover, fuel-efficient motorcycles in

 $<sup>^{1}</sup>$ In our data, motorcycles of engine size 125 cc have a fuel economy starting from 45 km/l (the Mahindra Rodeo 2016 model), ranging up to 78 km/l (the Yamaha Saluto 2016 model) and those of engine size 150cc have fuel economy ranging from 37 km/l (the Hartford VR 2018 model) to 72 km/l (the Bajaj Discover 2004 model).

<sup>&</sup>lt;sup>2</sup>Using the examples we provide in the previous footnote, for a person who drives 40 kms per day on average,

Nepal are not necessarily much more expensive than fuel-inefficient ones. Therefore, investing in fuel-efficiency can not only address the negative externalities due to pollution, but it can also provide significant private gains (in terms of fuel cost savings). Given that total lifetime costs (defined as the sum of purchase price and discounted operating costs over the lifespan of a vehicle) are likely to vary across different motorcycle models depending on their price and fuel efficiency, as well as on personal use, it is therefore plausible that not all individuals would be making the most optimal choices in terms of minimizing total lifetime costs.<sup>3</sup>

This finding resonates with a large literature on the energy-efficiency gap, which has shown that individuals fail to make optimal decisions in many situations when the benefits and costs of owning energy-consuming durables (such as motorcycles, or appliances) are unevenly distributed over time (Hausman, 1979; Jaffe and Stavins, 1994; Train, 1985). More energy-efficient durables, for example, generally have lower operating costs and thus may be less expensive to own over the long-run (even from a private perspective, without accounting for any underlying environmental or health benefits).

Reasons for the energy-efficiency gap in the literature have been ascribed to market failures as well as behavioral anomalies, that prevent individuals form realizing the gains from investing in energy-efficient technologies (Gillingham and Palmer, 2014).

The set of market failures that have been found to impede decisions to invest in energy-efficient technologies include asymmetric or imperfect information, principal-agent problems, unpriced externalities, as well as credit or liquidity constraints. Mullainathan and Thaler (2000) broadly categorize behavioral anomalies based on agents either displaying "bounded rationality", such as cognitive/skills limitations, framing problems, loss aversion, risk aversion and limited attention; or displaying "bounded willpower", such as myopia or present bias.

In this paper, we focus our attention on a set of behavioral anomalies, namely present bias, loss aversion and risk aversion. In particular, we are interested in assessing the impact that present bias may have on the types of motorcycles that individuals have purchased, based on their fuel economy. Given that the purchase of a vehicle is a large investment, and given that we are conducting our analysis in a low-income context, we hypothesize that time and risk preferences may contribute towards undermining the adoption of fuel-efficient motorcycles.

Present bias implies that individuals exhibit a high discount rate in the short run but a relatively low discount rate in the long run. Theoretically, this behavior has been modeled using a (quasi) hyperbolic time discounting function (Frederick et al. 2002; Laibson 2011). Individuals who are myopic place less weight on the future, possibly due to the uncertain utility that they would derive from future consumption (Strotz, 1955). This present bias may cause temporizing

opting for the less fuel efficient model would translate into needing an extra 137 litres of petrol annually (for 125 cc) or 192 litres annually (in the 150 cc category), to drive the same distance. In monetary terms, this could translate to savings in the range of Rs. 15'000 (USD 129.04) to Rs. 20'000 (USD 172.06) per year when using the most fuel efficient motorcycle of the same engine size (assuming an exchange rate of Nepal Rs.  $10,000 \approx \text{USD } 86.03$ , as of 29th March 2021). These calculations are based on motorcycle fuel economy specifications provided by the manufacturers, and the average fuel price in Nepal is assumed to be Rs. 110 per litre (as on January 26, 2021). The on-road fuel efficiency naturally also depends on road, traffic and weather conditions, as well as driving style and several other practical factors which are assumed to be the same for the purpose of this comparison.

<sup>3</sup>Other types of costs such as maintenance, financing, insurance and depreciation may also depend on the intensity of use, the purchase price as well as the lifespan of the motorcycle. For the purposes of the analysis in this paper, we only consider total lifetime costs in terms of operating (fuel) costs.

behavior on the part of individuals, especially when the costs are immediate. Present bias observed in situations where consumers have to incur a higher upfront cost for the purchase of an efficient technology has been found to play a role in exacerbating the energy-efficiency gap (Bradford et al., 2017; Cohen et al., 2017; Fuerst and Singh, 2018; Harding and Hsiaw, 2014). Thus, present-biased individuals may find it difficult to factor in future energy cost savings in a manner consistent with a rational agent economic model of discounting. Moreover, the considerable upfront effort required for estimating these savings can create a reluctance to exert this effort among present-biased individuals. This symbiosis between between present bias and effort was investigated by Augenblick et al. (2015), who found that even when there may be limited evidence of present bias over monetary payments, there may be significant present bias and thus dynamic inconsistency in choices over effort. In our context, this implies that present-biased individuals may tend to inflate the upfront effort required to do a lifetime cost calculation, and undervalue the future benefits of doing this calculation.

In a developing country setting, the possible role of present bias in determining investment decisions deserves even more attention from researchers. Individuals in developing countries are often exposed to risk, have limited social security options, and also have a reduced scope for borrowing in order to undertake investments. As present-biased individuals are likely to have high marginal propensities to consume, and thus have difficulties saving and maintaining liquid assets, it is likely to create additional liquidity constraints. This renders it difficult for these individuals to undertake investments, without additional provisions of credit (Kremer et al., 2019). Thus, while credit or liquidity constraints are a form of market failure that may prevent individuals from making investments in energy efficiency in themselves (particularly in cases when energy-efficient durables are more expensive), present bias in itself can introduce an additional source of liquidity constraints.

Our objective in this paper is twofold, a) to evaluate the determinants of behavioral anomalies that have previously been found to be relevant in explaining the energy-efficiency gap using survey data from Kathmandu, and b) to assess the impact of the presence of these anomalies on the energy-efficiency gap in the adoption of fuel-efficient motorcycles. We utilize novel data on existing petrol-driven motorcycle owners in Kathmandu for this purpose and construct a measure of the energy-efficiency gap, using which we intend to capture possibly inefficient choices made by individuals.

In order to address our first research question, we estimate a set of discrete-choice models to discern the socio-economic variables that may play a role in determining the behavioral anomalies. Our methodology in the second part of the analysis, where we are evaluating the role of these anomalies on a measure of the energy-efficiency gap, involves estimating linear models, sample-selection models as well as instrumental variable models employing a control function approach in order to correct for possible endogeneity of the present bias indicator.

We find that income is a strong determinant of time preferences, as well as risk and loss aversion in our sample, with relatively higher-earning individuals being less likely to be loss averse, more likely to be risk-averse, and more likely to have high discount rates. In addition, having a university education implies having lower discount rates, and being less risk-averse. Individuals who stated that they themselves, or a family member suffered from a respiratory disorder were more likely to be present-biased, and more likely to have high discount rates. We also find that certain psychological factors affected the occurrence of these anomalies; for instance, individuals who stated that they lived according to religious principles were more

likely to be loss-averse as well as risk-averse, as were those who stated that they held on to their family traditions. On the other hand, individuals who stated that they were likely to buy the latest technological innovation (even if they didn't need it) were more likely to be present-biased, as well as loss and risk-averse.

In the second part of our analysis, we find that present-biased individuals are more likely to have made relatively inefficient choices with respect to the fuel economy of the motorcycle that they purchased. They are more likely to have bought motorcycles with relatively larger total lifetime costs than the total lifetime cost that is achievable with buying the most fuel-efficient model of the same engine size. We do not find this effect for either loss or risk aversion. Moreover, while we find that individuals who applied for a loan to purchase their motorcycle were more likely to buy relatively fuel-inefficient vehicles, we also learn that individuals having higher levels of income have invested in relatively inefficient motorcycles (controlling for vehicle specifications such as engine size and brand). This effect is also echoed in respondents who own more than one motorcycle, i.e. they are also more likely to have made relatively fuel inefficient purchases for the vehicle that they use most frequently. Lastly, individuals whose families use the motorcycle regularly are more likely to have bought relatively fuel-inefficient motorcycles. We find that these results are confirmed across different specifications and assumptions.

These findings suggest that present bias is likely to be important in determining the fuel economy choices for motorcycles in Kathmandu; individuals who are present-biased or myopic are less likely to have realized the expected future cost savings from investing in fuel-efficiency. While it is difficult for us to pinpoint the exact channel through which present bias determines choices given our data, we hypothesize that these effects may either be driven by consumers undervaluing future fuel cost savings, or by the fact that present-biased individuals may find it onerous to compute total lifetime costs. Being cognizant of the total lifetime costs requires effort, not just in doing the computation, but also requires knowing how long one wants to own the motorcycle, the fuel efficiency of the motorcycle, the relevant discount rate to use, etc.

The finding on the effects of higher income suggests to us that it is not necessarily poorer individuals who purchase fuel-inefficient vehicles in this context. At the same time, our finding about individuals who applied for a loan making less efficient choices seems to suggest that being liquidity or credit-constrained may also lead to buying vehicles having relatively high total lifetime costs when fuel-efficient motorcycles are more expensive. While we do not have information in our data on whether these individuals were granted a loan or not, this finding suggests that they may also have chosen not to exert effort do lifetime cost calculations, or that they may have prioritized other attributes over fuel economy in choosing their motorcycles.

Our contribution to the literature is that to the best of our knowledge, this study is one of the first to evaluate the socio-economic determinants of these behavioral anomalies for Nepal, and one of the first to look at the role of psychological factors in explaining these anomalies in a developing country context. We construct a novel measure to capture the energy-efficiency gap in the purchase of vehicles, and complement this literature by evaluating the role of specific anomalies in determining fuel economy choices in a low-income country context.

Our results have important implications for the design of transport policy instruments in similar contexts: if present bias is likely to hinder the adoption of fuel-efficient vehicles, then it is plausible that information campaigns stressing the long-term benefits of switching to cleaner transport would be unlikely to succeed. On the other hand, the prevalence of present bias

also implies that standards as well as product taxes may be more relevant than fuel taxes in ensuring that individuals purchase more fuel-efficient vehicles. Currently, there are no emission standards for motorcycles in Kathmandu. Another effective policy option could be to relax liquidity constraints for households looking to buy fuel-efficient motorcycles (possibly through subsidies or low-interest loans). Economic incentives such as subsidies may also be useful for people who are loss or risk averse as they would lower the upfront costs of an investment.

The rest of the paper is organized as follows: Section 2 provides a review of the previous literature, Section 3 includes details on our data and the empirical strategies, Section 4 presents the main results, and Section 5 concludes.

#### 2 Previous Literature

In this section, we summarize the literature that is relevant to the discussion of this paper. This paper can fit into four strands of the economic literature. The first one focuses on the effects of various market failures and behavioral anomalies in determining the energy-efficiency gap. The second studies the barriers to technology adoption in developing countries, and in particular on the role of present bias, as well as of liquidity constraints. Our study also fits into the (third) strand of literature that evaluates the determinants of various behavioral anomalies. Lastly, we can contribute to a large literature in transport economics, in which some studies have evaluated the determinants of vehicle attribute choices (such as fuel economy).

The energy-efficiency gap has been attributed in the economic literature to market failures and behavioral anomalies that prevent individuals from adopting energy-efficient technologies which can both reduce operating costs, and environmental damage from the use of energy (Gillingham and Palmer 2014; Shogren and Taylor 2008). Examples of market failures include information imperfections and asymmetries (Allcott and Taubinsky 2015; Anderson and Newell 2004), environmental externalities, principal-agent problems and credit constraints or liquidity constraints (Golove and Eto, 1996). Behavioral anomalies, on the other hand, include myopia, loss and risk aversion, cognitive limitations, inattentiveness (or limited attention), as well as reference-dependent preferences (Gerarden et al., 2015).

Myopia has been found to undermine the adoption of energy-efficient technologies, at least in some contexts. For instance, Allcott and Wozny (2014) found that US consumers were indifferent between one dollar in discounted future petrol costs and 76 cents in vehicle purchase price, providing some evidence of discount rates playing a role in the adoption decision for more fuel-efficient cars. Cohen et al. (2017) used data on the refrigerator market in the UK and conclude that consumers exhibited moderate levels of myopia. Harding and Hsiaw (2014) found that present-biased individuals consumed more electricity than consumers who were not present-biased before joining a goal setting program. Interestingly, Bradford et al. (2017) found, using US data, that both discount factors as well as present bias are significantly associated with several kinds of activities in the health, finance and energy domains; for instance, they found that individuals who were not present-biased were much more likely to purchase fuel-efficient vehicles (in a similar strain to what we find in our study), whereas individuals having higher discount factors (and thus lower discount rates) were more likely to install energy-efficient lighting. The main reasoning provided in the paper for non-present-biased individuals to buy fuel-efficient cars is that cars qualify as "big-ticket" items, and it is likely that individuals may

face liquidity constraints when making their purchase, which renders time preferences to be potentially important. On the other hand, Sallee et al. (2016) used data from wholesale used car auctions to conclude that buyers fully valued fuel economy, i.e. that the prices of used cars moved one for one with changes in future discounted petrol costs. Busse et al. (2013) also found very low likelihoods for consumer myopia in the US new and used car markets, with implicit discount rates mostly near zero.

Present bias or myopia has traditionally been measured in terms of time inconsistency with respect to monetary rewards using a multiple price list approach, however it has also been conceptualized in terms of the choice to exert effort. As an example of the latter, Augenblick et al. (2015) show, using an experimental approach, that individuals may exhibit present bias in terms of effort tasks (such as text transcription, or solving puzzles and games), i.e. procrastinate, even if they are not present-biased in terms of monetary rewards. This definition of present bias has been elaborated as well as tested in several other studies, such as Berkouwer and Dean (2019); Carvalho et al. (2016); Dean and Sautmann (2021); Kaur et al. (2015); Lockwood (2020). In this paper, following the approach of other studies focusing on the effect of present bias on the energy-efficiency gap Bradford et al. (2017); Schleich et al. (2019), we measure present bias in terms of time inconsistency in monetary rewards, however we provide some suggestive evidence that individuals categorized as being present-biased by this method, may also be averse to exerting effort in order to undertake lifetime cost calculations.<sup>4</sup>

Loss aversion as well as risk aversion have also been found to be important determinants of the fuel economy choices of vehicles, as well as investment decisions in energy-efficient durables. Greene et al. (2013), for instance, found some evidence to suggest that loss aversion played a role in fuel economy decisions for individuals in the US, due to uncertainty about fuel economy as well as future fuel prices. Schleich et al. (2019) found, using data from a large-scale survey across eight European countries, that more risk-averse individuals, more loss-averse individuals and individuals who exhibited a lower time discount factor were less likely to adopt energy-efficient technologies (such as LED lamps, appliances, or retrofit measures). On the other hand, they found weaker evidence to suggest that present bias played a role in the choices of individuals (their results suggested that present bias affected choice only when they did not account for extreme values). Heutel (2019) also found that loss aversion may be relevant in explaining under-adoption of high-efficiency light bulbs, replacement of air conditioners, and investment in alternative fuel vehicles, using data from the US.

Our focus is on present bias, and the role that it plays on the adoption of durables having high initial costs, such as motorcycles, in a low-income setting. Present bias has been found to be an important deterrent towards many behaviors in developing countries: given that many individuals in these cases exhibit liquidity constraints, as well as high marginal propensities to consume, present bias is likely to make them seem even more impatient compared to richer individuals (Cassidy, 2018). Present bias in turn generates its own liquidity constraints (Angeletos et al., 2001), which implies that often in developing countries, it is difficult for present-biased agents to respond to surprise opportunities for investment without accompanying provisions for credit (Kremer et al., 2019). These liquidity constraints also cause households to decline high-return investments, such as in preventative health, or energy-efficient technologies.

Several studies have evaluated the determinants of behavioral anomalies, especially in the

<sup>&</sup>lt;sup>4</sup>The empirical analysis in Section 4 provides details.

psychology and behavioral economics literature. For instance, risk aversion has been found to be correlated with the presence of macro as well as micro-level shocks (Sakha, 2019) in Thailand, with gender, age, height, and parental background in Germany (Dohmen et al., 2011), and with household circumstances (such as success with agricultural operations, as well as income and assets) in Ethiopia (Yesuf and Bluffstone, 2009). The degree of loss aversion has been found to be influenced by socio-economic determinants such as age (positively) and education (negatively) by Hjorth and Fosgerau (2011) and Booij and van de Kuilen (2009). Income has also been found to be a determinant in certain contexts, as well as gender with some studies finding that females are more likely to be loss averse than males (Booij and van de Kuilen, 2009). The prevalence of present bias has been found to be higher in individuals with low levels of income in some studies (Can and Erdem, 2013), but not in others (Meier and Sprenger, 2010). In health-related contexts, present bias has also been found to be correlated with physical activity, i.e. individuals who exercised for longer periods were less likely to be present-biased (Hunter et al., 2018). Moreover, present bias has been found to be negatively correlated with age, and positively with education (i.e. more highly educated individuals were more likely to be present-biased), while gender has been found to be an insignificant determinant of present bias by Meier and Sprenger (2010), using data from the US. Thus, while some mixed evidence exists on the role of socio-economic factors in determining these behavioral anomalies, there is scant evidence from low and middle-income countries. We contribute to this literature, by using novel data from Nepal. Moreover, we also explore the role of some specific psychological factors in determining the presence of behavioral anomalies among respondents.

Lastly, this paper also fits into the transport literature on the determinants of fuel economy choices of individuals. Hedonic models, have been used for modeling vehicle model attributes, such as fuel economy and horsepower, as a function of the market price of the vehicle to estimate consumers' implicit willingness-to-pay for each individual feature (Espey and Nair, 2005). Alberini et al. (2019) found that more fuel-efficient variants of a car costed more using sales data on new cars from eight European countries, but that future fuel costs were heavily discounted by individuals, and that people expected very short payback periods on these models. Nayum and Klöckner (2014) found, using Norwegian survey data, that the intention to buy a fuel-efficient car, brand loyalty, number of cars owned and the number of drivers license holders in the household, as well as socio-economic factors such as household size, and income had significant direct effects on choosing a more fuel-efficient car. Using data on Swiss car buyers, Peters et al. (2011) used psychological theories such as the theory of planned behavior (TPB) and the norm activation model (NAM) to examine the determinants of fuel economy. They found that traits such as valence of less power and small size, general attitudes regarding fuel-efficient and small cars and perceived behavioral control were direct predictors of fuel economy choices. Our study intends to complement these studies, by explicitly modeling the energy-efficiency gap in the purchase of motorcycles in Nepal, and evaluating the role of behavioral anomalies in consumers not making the most efficient (or cost-minimizing) choices.

## 3 Data and Methodology

#### 3.1 Data

Our study is based on data drawn from a survey of a total of 2,500 respondents in the districts of Kathmandu, Bhaktapur, and Lalitpur, i.e. in the region of Kathmandu valley in Nepal. The field survey was conducted in the form of computer assisted personal interviews (CAPI) over three weeks during the months of October and November in 2019 in collaboration with a local survey partner.

The objective of this survey was to sample eligible individuals (older than 16 years, the legal age for obtaining a drivers license in Nepal) who were looking to purchase a motorcycle in the next few months, or were at least considering it. Thus, survey participants were sampled at places where one might expect a higher share of potential and existing motorcycle owners, such as at universities, near motorcycle dealerships, public and office parking places.

Out of the 2500 respondents sampled in the survey, 839 stated that they already owned a motorcycle. We were able to extract vehicle-related information (such as the brand, model name, engine size as well as on other technical specifications) based on the information that the respondents provided us for 668 respondents. In addition to vehicle specification-related information, we also collect socio-economic data from our respondents, as well as some psychological as well as behavioral information.

Given the large share of motorcycle ownership in the Kathmandu valley, as well as the opportunities that are available to potential buyers to invest in more efficient vehicles, this region was a natural choice of location for this study. At the start of the CAPI-based survey, enumerators informed the respondents about the goals of the study, conditions and incentives for participation, data privacy, and provided simple instructions on completing the questionnaire. The survey was first prepared in English by the research team and translated to Nepali by our survey partner prior to the field implementation. Participation in the survey was incentivized, with 20% of the respondents being eligible for a lottery.

In the first part of our analysis, we are interested in evaluating the socio-economic as well as psychological determinants of some behavioral anomalies. We use the entire data sample for conducting this analysis. In the second part, given that we are interested in the fuel economy choices of individuals (and possible prevalence of the energy-efficiency gap), i.e. we are interested in the intensive margin, we will be using data on the sub-sample of individuals who already own a motorcycle. However, in order to mitigate concerns of sample selection, we also estimate models using a Heckman-based approach in the second part of the analysis (refer to section 3.2 section for details).

Our focus in the second part of this study is on a specific set of behavioral anomalies that we hypothesize may be relevant in determining the energy-efficiency gap with respect to fuel-efficient vehicles in Kathmandu. These are present bias, loss aversion, as well as risk aversion. Present bias can be detected using either the multiple price list (MPL) approach, or choices over effort tasks. In this paper, we choose to test for present bias as well as capture time preferences using the former approach (which has been adopted in other studies in the energy economics literature), however we do not use this for eliciting parameters related to loss

aversion or risk aversion, and instead choose to adopt other methodologies from the literature.<sup>5</sup>

In Figures 1 and 2 in the Appendix, we provide the MPL-based questions that we used to measure present bias as well as time preferences in the survey. These questions are adapted from Schleich et al. (2019). In these price list-based questions, respondents faced a list of choices between two monetary options, A and B, and they were asked to pinpoint their preferred choice among these options. The monetary values were expressed in Nepali Rupees, and the responses were not incentivized, i.e. we did not pay the respondents based on their choices.<sup>6</sup>

Following Schleich et al. (2019), the MPL-based question consisted of two parts, each offering respondents seven choices. In the first part, respondents had to choose between Option A which yielded a monetary gain to be paid in one week, and Option B which specified an amount to be paid in 6 months, information which we use to determine the discount factor  $\delta$ . In the second part, Option A specified a possible monetary gain to be paid out in six months and one week, while option B denoted the amount that could be paid out in 12 months, which we use together with the first part to determine the present bias indicator  $\beta$ . The rationale for using this approach is that if a respondent chose Option A more frequently, the more she discounts future gains (i.e. the lower would be her discount factor). Present bias, on the other hand, can be detected by accounting for differences in choices of respondents between the two parts of this question, which would be a sign of a possible inconsistency in time preferences.

One point of difference with the methodology of Schleich et al. (2019) is that in calculating these two parameters, we assume a risk-neutral utility function, where the utility derived from a monetary payment is equal to the value of the payment. Schleich et al. (2019) solve for four preference parameters using four equations, allowing for risk aversion in the utility function, i.e. using a constant relative risk aversion (CRRA) utility function. Our choice to use the risk-neutral utility function largely stems from constraints related to the length of the survey. While we do not compute loss and risk aversion parameters using the MPL approach, we measure these anomalies using alternative approaches used in the literature (as described below), and thus are able to account for these anomalies in our estimations as well.

More details on how the preference parameters (namely the discount factor, as well as the present bias parameter) are calculated are provided in the Appendix.

We measure risk preferences by asking respondents to state how risk-loving they are, on a

<sup>&</sup>lt;sup>5</sup>We chose the MPL approach, because budgetary restrictions prevented us from doing in-person surveys multiple times, which is required for implementing effort tasks (Augenblick et al., 2015; Berkouwer and Dean, 2019). Budget considerations and other practical constraints also resulted in us prevented us from using the MPL approach to compute parameters for loss and risk aversion. Given the breadth of information that we were seeking to collect with the survey, its duration would have been very long had we adopted the MPL approach to assess these parameters. Prior to the baseline survey, we conducted a pilot test with 104 respondents, where we implemented a MPL approach to assess loss and risk aversion. The median duration of the survey was about 52 minutes, and many respondents expressed frustration at the length of the survey. Based on this feedback, we adapted the questions, as well as reduced the length of the survey. The final questionnaire took about 22 minutes to complete (median duration).

<sup>&</sup>lt;sup>6</sup>While incentivizing has been found to be important in the energy economics literature for respondents to honestly reveal their true preferences (Schleich et al., 2019), other economic studies have found that having to make hypothetical or unincentivized choices over money provides fairly similar results (Ubfal 2016, Falk et al. 2016). While we did not incentivize the responses due to budget-related constraints, we attempt to address any shortcomings that this may have created in our measures using an instrumental variable approach, as discussed in the next subsection.

scale of zero to ten, where choosing zero denotes someone completely unwilling to take risks, and choosing ten implies that the respondent is very willing to take risks. This method for capturing risk preferences has been adopted in several other studies (such as Ding et al. (2010); Dohmen et al. (2011); Hardeweg et al. (2013)). We convert this categorical variable to a dummy variable for our main analyses, by categorizing all individuals who stated that their score was less than 5 as being 'risk-averse'.<sup>7</sup>.

In order to capture possible loss aversion among our respondents, we adopt an approach suggested by Heutel (2019). We asked the respondents to choose between two motorcycles having different fuel economies. In the first part of this question, the respondents were told that they owned a 110 cc motorcycle, with annual fuel costs of Rs. 20,000, and they were offered the possibility to replace this with a 125cc motorcycle which had annual fuel costs of Rs. 30,000, i.e. they were asked whether or not they would like to replace their current vehicle with the larger one. The second part of this question involved assuming the opposite situation, i.e. respondents were told that they owned the bigger motorcycle having higher fuel costs, and they were asked whether they would replace it with the smaller one. Respondents who chose to keep the current motorcycle in both parts of the question (rather than replace it) are categorized as being loss averse.

Summary statistics on these parameters are provided in Table 1, along with those for the other explanatory variables that we use in our analysis. We report these statistics for the entire sample of respondents for whom we have non-missing information on socio-demographic variables (2245 observations) as well as for those respondents who own a motorcycle and for whom we have non-missing information on socio-demographic information as well as vehicle specifications (591 observations). These summary statistics are computed for the sample of respondents who are the main decision-maker with respect to the purchase of durables in their household, and who are the main users of their motorcycles. We also exclude observations in which the engine size of the motorcycle was greater than 500 cc.<sup>8</sup>

From Table 1, we find that the mean value of the present bias parameter  $\beta$  is about 0.999 for both the the entire sample, and for the sub-sample that already own a motorcycle, suggesting a very slight present bias averaged over the sample. About 12% of respondents are present-biased in both groups, while 50% of the sample is risk averse (the share is slightly higher among those who own a motorcycle), and about 7% can be categorized as being loss averse according to our measure. The mean annual discount factor for our sample  $\delta$  is about 0.87, which is slightly higher than the value 0.851 that Schleich et al. (2019) found for a sample of European countries.

The average age of respondents in our sample is about 29 years for the entire sample, and 30 years for those who own a motorcycle. About 29% of the whole sample is female, whereas this share shrinks to about 15% for existing motorcycle owners (among female respondents in our survey, about 15% already own a motorcycle, whereas the remaining 85% are looking to buy a new motorcycle). About 34% of the respondents in the whole sample have children,

<sup>&</sup>lt;sup>7</sup>In section 4.3, we estimate some models using alternative definitions of risk aversion, and find that our main results are robust to the choice of specification.

<sup>&</sup>lt;sup>8</sup>The majority of the motorcycles in Nepal belong to 100-180 cc engine sizes; we consider motorcycles above 350cc to be premium or luxury vehicles, moreover there are only two such observations in our data.

 $<sup>^9</sup>$ A  $\beta$  value equal to one denotes a person who is neither present-biased, nor future-biased,  $\beta>1$  denotes a future-biased individual, whereas  $\beta<1$  denotes a present-biased individual.

Table 1: Summary Statistics of Explanatory Variables

Sample		Entire sai				Sub-sample owning		
Explanatory variable	Mean	Standard Deviation	Minimum	Maximum	Mean	Standard Deviation	Minimum	Maximum
	Behavioral	anomalies and time p	references					
Present bias indicator	0.115	0.319	0	1	0.118	0.323	0	1
Beta	0.999	0.091	0.561	1.782	0.999	0.105	0.561	1.782
Annual discount factor	0.867	0.185	0.259	0.955	0.872	0.173	0.259	0.955
Whether risk averse	0.493	0.500	0	1	0.535	0.499	0	1
Whether loss averse	0.070	0.256	0	1	0.073	0.260	0	1
	S	ocio-economic variable	25					
Age of respondent	28.629	7.034	17	58	30.433	7.906	18	58
Whether female	0.287	0.453	0	1	0.147	0.355	0	1
Whether student	0.229	0.420	0	1	0.132	0.339	0	1
Whether have children	0.335	0.472	0	1	0.428	0.495	0	1
Monthly household income (shares)								
Less than Rs. 30,000	19.420				10.830			
Between Rs. 30,000 and 50,000	40.940				35.700			
Between Rs. 50,000 and 75,000	25.210				35.870			
More than Rs. 75,000	14.430				17.600			
Whether have a Bachelor's or Master's degree	0.502	0.500	0	1	0.492	0.500	0	1
Whether member of a club or society	0.299	0.458	0	1	0.347	0.476	0	1
Whether self/family member known to have a respiratory disorder	0.090	0.287	0	1	0.103	0.304	0	1
Whether receive remittances from family members abroad	0.205	0.404	0	1	0.208	0.406	0	1
Whether self/someone known owned a motorcycle during the blockade		0.459	0	1	0.717	0.451	0	1
Whether correctly solved math question on lifetime costs	0.211	0.408	0	1	0.242	0.429	0	1
Whether own more than one motorcycle					0.052	0.223	0	1
Whether applied for loan to purchase motorcycle					0.092	0.289	0	1
Whether motorcycle is second-hand					0.220	0.415	0	1
Whether family use motorcycle regularly					0.457	0.499	0	1
Whether aware of fuel economy when bought motorcycle					0.785	0.411	0	1
Whether reimbursed for petrol costs					0.234	0.423	0	1
	l	Psychological variables	;					
Whether live according to religious principles	0.720	0.449	0	1	0.741	0.438	0	1
Whether hold on to family's old traditions	0.771	0.421	0	1	0.799	0.401	0	1
Go out a lot (dinners, parties, other leisure activities etc.)	0.675	0.468	0	1	0.702	0.458	0	1
Likely to buy the latest technological innovation	0.548	0.498	0	1	0.548	0.498	0	1
Enjoy my life to the full	0.908	0.289	0	1	0.942	0.233	0	1

Note: The table reports the means, standard deviations as well as minimum and maximum values for the main explanatory variables. The relevant sample size is 2245 observations in columns (1) to (4), and 591 observations in columns (5) to (8) for whom we have information on vehicle specifications (except for the "whether applied for loan" variable, for which it is 590 observations due to a missing value). The sample comprises respondents who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample.

whereas this share is higher for the sub-sample owning a motorcycle, at 43%. About 50% of both samples have a university education, whereas the share of respondents who were students during the survey was relatively lower in the group that owns a motorcycle (at 13%, compared to 23% for the entire sample). Most of the respondents belong to households that earn a monthly income between Rs. 30,000 and Rs. 50,000 in the whole sample, whereas a majority belong to the group that earns between Rs.50,000 and Rs.75,000 in the sub-sample owning a motorcycle, i.e. motorcycle owners earn higher incomes in our sample (as expected).

About 21% of respondents in the whole sample as well as the sub-sample receive remittances from family members living abroad, while about 9-10% of the respondents know that either they or their family members have a respiratory disorder in both samples. About 30% of respondents stated that they were a member of some club or group (such as a neighborhood associations, student associations, sports groups, etc.) in the entire sample, with the share being marginally higher at about 35% in the sub-sample. We also asked a simple multiple-choice question in the survey to assess the computational skills of the respondents; having provided them with the purchase price, expected duration of ownership, and annual operating costs, we asked them to compute the total lifetime costs (assuming zero discounting). We find that only about 21% of respondents answered this question correctly in the whole sample. This share was not much higher among those who owned a motorcycle, at about 24%. Lastly, about 70% of respondents in both groups stated that either they, or someone they knew, owned a motorcycle during the blockade of 2015. 11

Among the sample that owned a motorcycle, we find that about 22% of respondents own a second-hand vehicle. 9% of respondents stated that they had applied for a loan in order to purchase their motorcycle, while about 5% stated that they owned more than one motorcycle. About 46% of respondents stated that other members of their family also used the motorcycle on a regular basis. A relatively high share (about 79%) of respondents stated that they were aware of the fuel economy of the motorcycle back when they bought it, while about 23% of respondents have their petrol costs reimbursed by an employer, or their family (if they are students, for example).

We also collected information on some psychological variables in our data. The five psychological variables in our data were measured as categorical variables, with four possible responses: whether the respective trait applied fully to the respondent, applied somewhat to the respondent, did not apply to them, or did not apply at all to them. In order to obtain clearer distinction between categories of responses, we converted these variables into dummy variables, with the base category including individuals who stated that it did not apply to them, or did not apply at all to them (and the variables taking the value '1' if they stated that it applied either somewhat, or fully, to them). We find that for about 72% of respondents, 'living according to religious principles' was a trait that applied to them, while about 77% stated that they are likely to hold on to their family's old traditions, in the entire sample. About 68% of respondents stated that

<sup>&</sup>lt;sup>10</sup>The exact question was: Suppose you buy a bike for Rs 300,000. Your annual cost of petrol is Rs. 20,000. You expect to use the bike for 10 years (lifetime of the bike). What would be the total cost over the lifetime of the bike? Assume that average cost of fuel, fuel economy, distance driven per year, expected lifetime of the bike remains the same, and assume that the value of Rs 1 today is the same as Rs 1 tomorrow. The possible responses were Rs. 450,000, Rs.500,00, Rs.600,000, Rs. 750,000 and "don't know", with Rs.500,000 being the correct answer.

<sup>&</sup>lt;sup>11</sup>The blockade was an economic crisis that arose due to ethnic tensions, and resulted in shortage of essential supplies such as fuel and medicines in Nepal.

they went out a lot (for e.g., for dinners, parties or other leisure activities), whereas about 91% of the respondents stated that the statement 'enjoy my life to the full' applies at least somewhat to them. Lastly, we also find that about 55% of respondents stated that they are somewhat likely to buy the latest technological innovations, even if they may not necessarily need them.

We provide additional information on the motorcycles owned in Table 5 in the Appendix. This includes information on the engine size, the year of purchase of the motorcycle as well as average fuel economy by engine size and year of purchase. The engine size as well as fuel economy represent the vehicle specifications as advertised by the manufacturers. We find that in general, larger motorcycles (in terms of engine size) are less fuel-efficient, although this is not a monotonic trend. The largest motorcycles in our data sample have a 350 cc engine capacity, whereas the smallest have a 100 cc size. Also, we observe that motorcycles which were bought earlier had higher levels of fuel economy, compared to those that were bought relatively recently. The oldest motorcycles in our sample are about 11 years old.

#### 3.2 Methodology

In this subsection, we explain the econometric approach that we adopt in this study. In the first part, we are interested in understanding the role of socio-demographic as well as psychological factors in determining three specific behavioral anomalies, as well as the time preferences for individuals in Kathmandu. Our second objective is then to evaluate the role of these behavioral anomalies in determining the energy-efficiency gap in the purchase decision for motorcycles.

The model that we estimate in the first part takes the following form:

$$A_i = \alpha_0 + \alpha_1 S_i + \alpha_2 P_i + \mu_i \tag{1}$$

In Eq.(1),  $A_i$  could represent one of four possible dependent variables; three dummy variables capturing the existence of specific behavioral anomalies (present bias, loss aversion and risk aversion), or the annual discount factor of the  $i^{\rm th}$  respondent.  $S_i$  denotes the set of socioeconomic explanatory variables, while  $P_i$  denotes the set of psychological determinants, and  $\mu_i$  denotes the residual.

In order to estimate the models that use the dummy variables denoting presence of behavioral anomalies as dependent variables, we use a probit methodology. For the model with the annual discount factor, given that the variable takes values between 0 and 1, we estimate a fractional response model, where we assume that the functional form of the annual discount factor is a cumulative normal density, i.e. we estimate a fractional probit model. As mentioned in the previous sub-section, these models are estimated for the entire sample of respondents.

In the second part of the paper, we estimate the effects of the three behavioral anomaly indicators at the centre of our analysis as well as that of the discount factor, on the energy-efficiency gap, focusing the analysis on the sub-sample that already own a motorcycle. Of course, as discussed in Section 1, an energy-efficiency gap may exist due to the effects of other behavioral anomalies as well, such as limited attention, as well as cognitive limitations that impede lifetime cost calculation. As an example of the former, individuals may not pay attention to the fuel economy when they purchase their motorcycle (if fuel economy were not

clearly advertised, or if they simply chose to focus their attention on other attributes). In the latter case, consumers may not be able to evaluate the present discounted costs of buying fuel when they make their choice, due to the cognitive load that it necessitates.

Consumers observe various attributes of the product (such as engine power, brand, color, design, etc.), however they may have only a partial understanding of lifetime fuel costs (Sallee, 2014; Turrentine and Kurani, 2007). In order to do a lifetime cost computation, consumers need information on how much they are likely to drive, relevant discount factors, future fuel prices, as well as how long they expect to use the vehicle. This implies that there are considerable cognitive costs (as well as time and information-collection costs) in doing this computation. These costs may thus create, in some consumers, an aversion to exert upfront effort to do the calculation and identify vehicles have low lifetime costs, and this may be, as discussed above, representative of a form of present bias. When such costs are considerable due to present bias, the consumer may put in less effort than their "long-run self" (Lockwood, 2020). As a result, not all individuals would be making the most optimal choices in terms of minimizing total lifetime costs. In order to better distil the effect of possible present bias on the energy-efficiency gap net of the other anomalies that may confound this effect, we control for the possible limited attention paid by respondents to fuel economy, as well as for their computational ability in all our regression results.

We assume in this paper that individuals first decide the engine size of the motorcycle that they would like, and then make their final choice based on other factors (among them, possibly fuel economy), conditional on purchasing a vehicle of their preferred engine size. This sequential decision-making process, where individuals first decide the engine size of vehicle, enables us to quantify the extent of the energy-efficiency gap. Based on the discussion above, some consumers will end up purchasing relatively fuel-inefficient motorcycles in a given engine size category, resulting in an energy-efficiency gap with respect to the most fuel-efficient motorcycle model in that category.

For the estimations in the second part, we need to define the level of the energy-efficiency gap, i.e. our dependent variable. To do this, we first calculate the total lifetime cost of owning the motorcycle for each individual in our sample. The total lifetime cost of a motorcycle for respondent i is denoted as  $TC_i$  and is defined as:

$$TC_i = P_i^I + \sum_{t=0}^T \frac{1}{(1+r)^t} \frac{P_t^E D_i}{F_i}$$
 (2)

where  $P_i^I$  denotes the purchase price of the motorcycle in Nepali Rupees (Rs.),  $P_t^E$  denotes the per litre price of petrol in year t (about Rs. 110 per litre during the time of our survey),  $D_i$  denotes the average annual distance driven by respondent i in kilometres, and  $F_i$  denotes the fuel economy of the motorcycle. T denotes the lifespan of the motorcycle in years, and r is the discount rate. In calculating the lifetime costs in our main specifications, we assume a lifespan of 10 years for all motorcycles, i.e. T=10 (which represents the  $99^{th}$  percentile of the distribution of vehicle age in our data), and that the fuel price and fuel efficiency remains constant over the lifespan. Moreover, we simplify the consideration of costs in this study, by only computing the total lifetime costs in terms of operating (fuel) costs.

 $<sup>^{12}</sup>$ In section 4.3, we also estimate some models using a lifespan of five years, and find that our main results are robust to the choice of specification.

Next, for each engine size, we identify the motorcycle having the highest fuel efficiency; we then compute the lifetime costs using the fuel economy as well as purchase price of this model, while still using the actual distance driven by the individual ' $D_i$ ' for the calculation. This measure of total lifetime costs is used as a benchmark against which to compare the actual lifetime costs of the vehicle owned by the respondent, and to compute the energy-efficiency gap. The level of the energy-efficiency gap, and our dependent variable for respondent 'i'  $EE_i$ , is defined as the log of the ratio of the total lifetime cost of owning the motorcycle  $TC_i$ , to the total lifetime cost of the most fuel-efficient motorcycle of the same engine size  $TC_{min}$ , as shown below:

$$EE_i = log(\frac{TC_i}{TC_{min}}) \tag{3}$$

The value of the discount rate r to be considered in Eq.(2) is not always obvious. The discount rate not only captures the time value and opportunity cost of money, but it can also be thought of as the cost of borrowing. In a first variant of the outcome variable, we assume that T=10 and r=7%, i.e we assume that the discount rate equals the market interest rate. On the other hand, the second variant of the dependent variable sets the discount rate r equal to the average loan rate for motorcycle loans, at about 10%. As robustness checks, we also use alternative versions of the discount rate; these results are presented in columns (1) to (3) of Table 7.

Thus, higher values would imply that respondent 'i' made a relatively inefficient choice of motorcycle (since the total lifetime costs of the actual choice would then be higher than the total lifetime costs possible by buying the most fuel-efficient motorcycle). The model that we estimate, using these two variants of our dependent variable, can be expressed as follows:

$$EE_i = \beta_0 + \beta_1 A_i + \beta_2 X_i + \eta_i \tag{4}$$

where  $EE_i$  could denote either of the two versions of the dependent variable that we construct to represent the energy-efficiency gap in the choice of motorcycle of respondent i,  $A_i$  is the set of behavioral anomalies as well as the individual-specific discount factor,  $X_i$  denotes a set of socio-economic variables, and  $\eta_i$  denotes the residual.

We estimate this model using four methodologies. The first is the ordinary least squares (OLS) approach. The second is a sample-selection model (following Heckman (1976)), where the first stage models the decision of respondents to own a motorcycle. The third and fourth models adopt an instrumental variable control function estimation (following Wooldridge (2010)).

While our focus in the second part of the analysis is on the sub-sample of individuals who already own a motorcycle, we estimate a sample-selection model to mitigate concerns related to sample selection. This model is estimated using the maximum likelihood approach. In the selection equation, we include two variables that may influence the decision to own a motorcycle (without directly affecting the energy efficiency gap): the first is the number of bus stops

 $<sup>^{13}</sup>$ The average interest rate from 2010-2019 in Nepal was about 7.2% (Trading Economics, 2020).

<sup>&</sup>lt;sup>14</sup>Many banks in Nepal offer special loans for motorcycle purchases, independent of auto loans that are valid for car purchases. For instance, Everest Bank introduced a motorcycle loan in 2020, offering an interest rate of 9.98% (Nepal Drives, 2020).

within 500 metres of the centre of the locality where the respondent resides<sup>15</sup>, and the second is the number of motorcycle showrooms and shops within 500 metres of the centre of the locality where the respondent lives<sup>16</sup>. These variables intend to capture access to public transport, as well as the ease of finding and purchasing a new motorcycle, that are both likely to affect the decision to buy a motorcycle (but unlikely to directly influence the energy-efficiency gap, in our opinion).

In order to elicit the time preference parameters, namely the discount factor as well as an indicator for present bias, we use the MPL approach that we explained in Section 3.1. However, for the reasons that we discussed, it is possible that these variables are measured with error. This may engender endogeneity in our estimation. In order to address endogeneity due to measurement error, we additionally estimate instrumental variable (IV)-based models as the third and fourth methodological approaches. We treat the present bias indicator to be endogenous in these estimations.<sup>17</sup>

Given that the present bias indicator is a dummy variable, we estimate a control function-based model, following Wooldridge (2010). The use of the control function approach involves estimating a first stage probit model, where the present bias indicator is regressed on an IV as well as on all explanatory variables from the second stage estimation. The reduced-form residuals from this first stage estimation are then used as an instrument in a two-stage least squares estimation, with the relevant energy-efficiency gap measure as a dependent variable. The second-stage coefficients from this model serve as the final estimates of the control function approach.

In the first control function-based estimation in this paper, we use self-reported information on whether the respondents have experienced and suffered major damages such as death, injury or damage to property from natural disasters (including earthquakes, flooding and droughts) as an IV. In the second model, we continue to use this subjective measure as an IV, and additionally include a more objective IV measure capturing the intensity of the 2015 Nepal earthquake in the locality where the respondent lives. The earthquake intensity is measured in terms of the Modified Mercalli Intensity (or MMI) scale.

Natural disasters have been found to be important determinants of behavioral anomalies in the economic literature. Individuals who have experienced natural disasters such as floods and earthquakes have been found to be more likely to be present-biased, and discount the future more (Cassar et al., 2017; Sawada et al., 2015) to be more loss averse (Shupp et al., 2017) and also likely to be more risk averse (Cameron and Shah, 2015), however other studies also find that natural disasters could have the opposite effect (for instance, Shupp et al. (2017) find that after a tornado struck Oklahoma City in 2013, survey participants who lost a friend or neighbor were less risk averse.) Based on this evidence, exposure to natural disasters is likely

<sup>&</sup>lt;sup>15</sup>The centre of each locality was identified by the centroid of the polygon corresponding to that locality using Google Earth or Google Maps.

 $<sup>^{16}</sup>$ The showrooms refer to shops that sell new, as well as used motorcycles. These may be single-brand or multi-brand outlets, and may also include service and repair centres. The centre of each locality was identified by the centroid of the polygon corresponding to that locality using Google Earth or Google Maps.

<sup>&</sup>lt;sup>17</sup>Given that the annual discount factor was also derived using the MPL approach, it is likely that this variable is also measured with error. This implies that we could treat both variables as endogenous. However, it is not straightforward to estimate, as well as to interpret the results of, control function-based IV models with more than one endogenous variable. Moreover, our focus in this study is on the effect of present bias on the energy-efficiency gap. Thus, we choose to treat only this variable as endogenous in our estimations.

to influence each of the behavioral anomaly variables (including present bias) as well as the time preferences. Thus, by controlling for the discount factor and loss as well as risk aversion in our second stage estimation, we strengthen the argument for the exclusion restriction to be satisfied. Moreover, we choose to adopt both an individual-specific instrument on whether respondents have been affected by natural disasters (in our third model specification), as well a more objective measure of the intensity of a large-scale disaster where the respondent resides in the fourth specification (along with the subjective measure), so as to capture the effect of natural disasters on present bias in a more balanced and comprehensive manner.

As controls, we include several respondent-specific socio-economic variables such as income, education, gender, age, whether the respondent has children, etc. In addition, we include controls related to the motorcycle, as well as the respondent's use of it (such as whether it was second-hand, whether the respondent owns more than one motorcycle, whether the respondent needed to apply for a loan to purchase their motorcycle, and whether the family of the respondent also used the vehicle regularly). Moreover we also include controls for the engine size, the brand of the motorcycle, and the age of the motorcycle, along with the distance of the centre of the locality where the respondent lives from the city centre. We cluster standard errors by locality in all regressions.

#### 4 Results

In this section, we present the main results of the analysis. Table 2 includes the estimations of the socio-demographic as well as psychological determinants of the three behavioral anomalies that we consider in our study (present bias, loss aversion, as well as risk aversion) along with those of the annual discount factor. In Table 3, we then estimate the effect of each of these biases as well as time preferences on the energy-efficiency gap, using the two variants of the dependent variable that we defined in the previous section.

#### 4.1 Results on the determinants of behavioral anomalies

Columns (1), (2), (3) and (4) of Table 2 present the results of the coefficients of estimating the models on the determinants of present bias, the annual discount factor, loss aversion, and risk aversion, respectively for the entire sample of respondents. The respective marginal effects are presented in Table 6 in the appendix. The results of columns (1), (3) and (4) are estimated using a probit model, whereas the model of column (2) is estimated using a fractional response methodology. The reference category for the explanatory variable on monthly household income is individuals whose household income is less than Rs. 30,000, or 257 USD per month, whereas for the 'university' variable, it is individuals whose highest level of educational attainment is either high school or below or those having a professional/vocational education. The reference categories for the five psychological variables in these models are whether the respondent answered that the respective trait either did not apply to them, or did not apply at all to them.

In column (1) of Table 2, we find that respondents who stated that they receive remittances from family members are more likely to be present-biased. While receiving remittances may relieve liquidity constraints at least partially, it does not obviate the fact that such respondents

Table 2: Determinants of Behavioral Anomalies and Time Preferences

Dependent variable Model	Present biased Probit	Annual discount factor Fractional response	Loss averse Probit	Risk averse Probit
Socio-demo <sub>i</sub>	graphic variables			
Monthly household income				
Between Rs. 30,000 and 50,000	0.108	-0.062	-0.292***	0.410***
	(0.127)	(0.057)	(0.106)	(0.101)
Between Rs. 50,000 and 75,000	0.167	-0.028	-0.282*	0.425***
	(0.165)	(0.068)	(0.161)	(0.167)
More than Rs. 75,000	-0.115	-0.217***	-0.282**	0.580***
	(0.125)	(0.072)	(0.148)	(0.151)
Whether have a university degree (Bachelor's or Master's)	-0.011	0.130***	0.005	-0.292***
	(0.105)	(0.036)	(0.116)	(0.057)
Whether have children	0.109	0.066	-0.095	-0.005
	(0.109)	(0.056)	(0.127)	(0.100)
Whether student	-0.102	0.133***	0.009	0.008
	(0.113)	(0.056)	(0.109)	(0.078)
Age	0.009	0.001	0.012	-0.002
	(0.006)	(0.003)	(800.0)	(0.005)
Whether female	-0.058	-0.036	0.047	0.047
	(0.084)	(0.040)	(0.114)	(0.076)
Whether receive remittances	0.182**	0.074*	-0.166	-0.150**
	(0.085)	(0.042)	(0.156)	(0.080)
Whether have a respiratory disorder (self or close family)	0.386***	-0.182***	0.051	-0.014
, , , , , , , , , , , , , , , , , , , ,	(0.134)	(0.070)	(0.147)	(0.111)
Whether member of a club or society	0.322***	0.121***	-0.271***	-0.012
,	(0.115)	(0.047)	(0.115)	(0.111)
Whether self/someone known owned a vehicle during blockade	-0.226***	-0.210***	-0.137	0.042
<b>3</b>	(0.086)	(0.056)	(0.119)	(0.079)
Log of distance of locality from the city centre	-0.008	-0.007***	0.028	0.017**
	(0.006)	(0.003)	(0.018)	(0.008)
Psycholog	gical variables			
Live according to religious principles	0.159	0.028	0.391***	0.542***
5 · · 6 · · · · · · · · ·	(0.108)	(0.042)	(0.117)	(0.077)
Hold on to my family's traditions	0.186**	-0.127***	0.519***	0.153**
, . ,	(0.099)	(0.052)	(0.147)	(0.079)
Go out a lot (e.g. dinners, parties, other leisure activities etc.)	0.002	0.086	-0.041	-0.336***
(3	(0.110)	(0.056)	(0.139)	(0.106)
Enjoy my life to the full	0.140	-0.208***	0.244	0.357**
3.9 9 ×	(0.143)	(0.072)	(0.169)	(0.165)
Likely to buy the latest technological innovation	0.215**	0.027	0.367***	0.311***
. , ,	(0.110)	(0.048)	(0.111)	(0.096)
Observations	2245	2245	2245	2245

Note: The table reports the coefficients as well as standard errors in parentheses for the main explanatory variables. The sample includes both current motorcycle owners as well as respondents who don't currently own a motorcycle, and comprises respondents who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample. All models include dummies for districts. Standard errors are clustered at the locality level. \*,\*\* and \*\*\* respectively denote significance at 10%, 5% and 1% levels. Coefficient on constant is not reported.

may have been more cash-strapped to begin with, and that implies that credit or liquidity constraints might effectively play a role in determining present bias as well. Interestingly, we find that individuals who have a respiratory disorder, or know someone in their household who does, are also more likely to be present-biased, as are those individuals who are members of certain clubs or groups. We do not find any significant differences across gender, age, income or education, in determining present bias in our sample.

As far as the role of the psychological variables is concerned, we find that for individuals who stated that holding on to their family's traditions was important to them were more likely to be present-biased, than those who said that it did not apply to them. Intuitively, buying the latest innovation/gadget, even if it was not needed, was also positively associated with being present-biased in our sample.

In the model of column (2), we estimate the determinants of the annual discount factors of individuals in our sample. We find that individuals having high levels of income are more likely to have relatively low discount factors (and thus high discount rates) compared to lowerincome individuals (the indicator for household monthly income being higher than Rs.75,000 or 642 USD is significant at the 1% level). This is an interesting finding, as it suggests that richer individuals are more likely to discount the future more than relatively poorer individuals. Intuitively, we find that respondents having a university education are likely to have higher discount factors (and thus lower discount rates) than individuals who are only educated up to high school or below or those that have a professional education. This result is also valid for individuals who are currently students; students are more likely to have lower discount rates that non-students. The presence of respiratory disorders is negatively associated to the discount factor, i.e. individuals who know that they or their family members suffer from a respiratory disorder are likely to have high discount rates. With respect to the psychological variables, we find that individuals for whom holding on to family traditions is important are likely to have lower discount factors/higher discount rates, as do those individuals who stated that they enjoyed their life to the full.

In column (3), we report the results on the determinants of loss aversion among respondents in our sample. In line with intuition, individuals having higher levels of income are less likely to be loss averse, as the coefficients on the income category variables suggest. We find that individuals who were members of clubs or groups are also less likely to be loss averse, suggesting a possible role for more social ties in mitigating loss aversion. Among the psychological variables, we find some interesting correlations as well: for instance, individuals who state that they live by religious principles, hold on to their family's traditions, or like to buy the latest innovations (even if they don't need them) are more likely to be loss averse, than individuals for whom these qualities are not important (these coefficients are significant at the 1% level). These findings hint at the possible role of family traditions and culture in determining these biases.

Lastly, in column (4), we provide the results of the model estimating the determinants of risk aversion among respondents. We find that higher income has a significant and positive effect on risk aversion; each of the three dummy variables have a positive coefficient, and are significant at the 1% level. Furthermore, individuals who have a university education are less likely to be risk averse, compared to individuals who have lower levels of education. Receiving remittances also mitigates risk aversion partially; individuals who receive remittances are less likely to be risk averse than those who don't. Among the psychological variables, we find that each of the explanatory variables has a significant effect on risk aversion in our sample, and for

most of these variables, if the statement applies, individuals are more likely to be risk-averse, i.e. the coefficients are positive. The exception is the variable for going out a lot; individuals who stated that they went out a lot were less likely to be risk-averse than those who stated that going out a lot was not something they did, or was not applicable to them.

### 4.2 Results on the energy-efficiency gap

In Table 3, we present the results of estimations evaluating the effect of these anomalies on the energy-efficiency gap. As mentioned in the previous section, we use two dependent variables for this analysis. These are the log of the total lifetime cost of the owned motorcycle to that of the most efficient motorcycle of the same engine size, computed assuming a discount rate equal to the market interest rate of 7%, and assuming a discount rate equal to 10%. We estimate four models using each dependent variable; an OLS model, a Heckman-based approach to account for sample selection, an IV based model using the Wooldridge (2010) control function approach with a single instrument, and the Wooldridge model using two instruments. We use self-reported information on having been affected by natural disasters as an instrumental variable for the present bias indicator in columns (3) and (7), whereas in columns (4) and (8), we supplement this measure with a more objective measure, the MMI scale measure of the intensity of the 2015 earthquake for the locality where the respondent lives. A point to note about the IV-based control function estimation results in columns (3), (4), (7) and (8) of Table 3 is that the first-stage F-statistics are reasonably strong across estimations. Moreover, the coefficients are also of a similar order of magnitude across estimations.

Our main results suggest that present-biased consumers were associated with having a relatively higher ratio of total lifetime costs, compared to what they could purchase given the engine size of the vehicle, and given the distance that they drove on their motorcycle. We observe that the present bias indicator has a positive coefficient across all models, the OLS results of columns (1) and (5), in the Heckman results of columns (2) and (6) when we correct for sample selection, as well as in the IV-based results of columns (3), (4), (7) and (8) where we attempt to address for possible endogeneity of the present bias variable due to possible measurement error. It is reassuring for us to note that the coefficients on the present bias indicator are similar in columns (1) and (2), as well as in (5) and (6); these results suggest that sample selection may not be a significant concern in our estimations. This is further confirmed by the results of the Wald test for the independence of equations; in both columns (2) and (6), the null hypothesis that the selection equation and the main outcome equation are independent cannot be rejected even at the 10% level.

One might conjecture that the results of the effect of present bias on the energy-efficiency gap may be driven by whether or not individuals paid attention to fuel economy when they bought their motorcycles, i.e. it may be the outcome of limited attention. Indeed, we find that about 71% of present-biased individuals in the regression sample of columns (3), (4), (7) and (8) of Table 3 were aware of the fuel economy of their motorcycle when they bought it, compared to 80% of non-present-biased individuals (and this difference is significant at the 5% level, using a one-sided T-test and at the 10% level using a two-sided T-test). In order to mitigate this as a possible channel, we control for whether or not the respondent was aware of the fuel economy of the motorcycle when they purchased it. While this variable is insignificant across all specifications, it allows us to control at least partially for possible the effects of limited

Table 3: Energy-efficiency gap estimations

- International Control of Contro		j	( - +	( - 400. 400.000)		, Ja 20	(-+ +	(-4
Dependent variable Methodology	OLS	Log or ra Heckman	Wooldridge (2010): one IV	discount rate) Wooldridge (2010): two IVs	OLS	Log or rat Heckman	tio of total lifetime cost (10%) Wooldridge (2010): one IV	discount rate) Wooldridge (2010): two IVs
	(1)		(3)	(4)	(5)		(2)	(8)
Present bias indicator	0.021*	0.023**	0.070**	0.066*	0.021*	0.022**	0.071**	0.066*
-	(0.012)	(0.012)	(0.036)	(0.037)	(0.012)	(0.012)	(0.040)	(0.037)
Annual discount factor	0.026	0.023	0.017	0.018	0.025	0.023	0.017	0.018
Whether risk averse	-0.0004	-0.001	-0.006	900:0-	-0.0005	-0.001	-0.006	90:0-
	(0.007)	(0.007)	(0.009)	(0.009)	(0.007)	(0.007)	(0.009)	(0.00)
Whether loss averse	-0.001	-0.001	-0.005	-0.005	-0.001	-0.001	-0.005	-0.005
	(0.016)	(0.016)	(0.016)	(0.017)	(0.016)	(0.016)	(0.016)	(0.017)
Age of respondent	-0.0004	-0.0004	-0.0005	-0.0005	-0.0004	-0.0005	-0.0009)	-0.0009)
Whether have children	-0.007	-0.008	-0.009	-0.009	-0.006	-0.008	-0.009	-0.009
	(0.012)	(0.011)	(0.012)	(0.012)	(0.012)	(0.011)	(0.012)	(0.012)
Monthly household income								
Between Rs. 30,000 and 50,000	0.019**	0.016	0.023***	0.022***	0.018**	0.015	0.022***	0.022**
	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)
Between Rs. 5U,000 and 75,000	0.038	0.030***	0.043****	0.042****	0.038****	0.030"""	0.043****	0.042****
More than Rs. 75,000	0.042***	0.035***	(0.013) 0.049***	0.048***	0.041***	0.034***	(0.013) 0.049***	0.048***
	(0.012)	(0.012)	(0.013)	(0.013)	(0.011)	(0.012)	(0.013)	(0.013)
Whether student	-0.003	-0.0004	-0.001	-0.002	-0.004	-0.001	-0.002	-0.002
Whether female	-0.005	0.002	-0.002	-0.002	-0.006	0.002	-0.002	-0.003
	(0.000)	(0.011)	(0.009)	(0.009)	(0.00)	(0.011)	(0.009)	(0.009)
Whether have a university degree	-0.004	-0.004	-0.004	-0.004	-0.004	-0.004	-0.004	-0.004
Whether receives remittances	(0.008)	(0.008)	(0.009)	(0.009)	(0.008)	(0.008)	(0.009)	(0.009)
	(0.010)	(0.00)	(0.010)	(0.010)	(0.010)	(0.00)	(0.010)	(0.00)
Whether member of a club or society	-0.011	-0.011	-0.014**	-0.014***	-0.011	-0.011	-0.014***	-0.014***
Whether celf/family member known to have a recniratory disorder	(0.007)	(0.007)	(0.006)	(0.006)	(0.007)	(0.007)	(0.006)	(0.006)
whether sell/lailing member allowin to have a respiratory disolder	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)
Whether aware of fuel economy when bought motorcycle	0.002	0.002	0.005	0.004	0.002	0.002	0.004	0.004
	(0.011)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)
Whether correctly solved math question on lifetime costs	-0.006	-0.006	-0.009	-0.009	(0.010)	-0.006	-0.009	-0.009
Whether reimbursed for petrol costs	-0.002	-0.002	-0.006	-0.006	-0.002	-0.002	-0.006	-0.006
Whathar family was matarabala	(0.012)	(0.012)	(0.013)	(0.013)	(0.012)	(0.012)	(0.013)	(0.013)
Wileties family uses motolicycle regularly	(0.007)	(0.000)	(0.007)	(0.007)	(0.007)	(0.000)	(0.007)	(0.007)
Whether applied for loan to purchase motorcycle	0.032**	0.032***	0.037***	0.036***	0.032***	0.032***	0.037***	0.037***
	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)
Whether motorcycle is second-hand	0.014	0.014	0.013	0.013	0.013	0.013	0.012	0.012
Whother animal enemand anomal animal alumination	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)
vinetifel sen/someone known owned a venicle duming biochade	(0.004)	(0.006)	0.000	(2000)	(0.004)	(0.006)	0.008	30.00
Whether own more than one motorcycle	0.048**	0.048***	0.049***	0.049***	0.047***	0.047***	0.048***	0.048***
	(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.013)	(0.014)	(0.014)
Observations	579	2244	559	559	579	2244	559	559
Wald test for independence of equations		1.26				1.31		
P-value		0.261				0.2523		
Cragg-Donald F-statistic			62.386	62.838			62.386	62.838

Note: The table reports the coefficients as well as standard errors (in parentheses) for the main explanatory variables. The sample comprises respondents for whom vehicle specification data was available i.e., who own a motorcycle, and who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles in columns (1), (3), (4), (5), (7) and (8). In columns (6), we also include respondents who don't currently own a motorcycle. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample. All models include controls for district, as well as for brand, distance to the city centre, engine size and age of the motorcycle. Standard errors are clustered at the locality level: ","\* and "\*\* respectively denote significance at 10%, 5% and 1% levels. Coefficient on constant is not reported.

attention, and thus helps isolate the effect of present bias on the energy-efficiency gap.

Similarly, in order to control for cognitive limitations that inhibit individuals from using computational skills to calculate lifetime costs, we include a regressor measuring whether the individual correctly answered a math question on the computation of lifetime costs. We find that this variable is also insignificant across models in determining the energy-efficiency gap. Moreover, we find that about 24% of non-present-biased individuals answered this question correctly, whereas 35% of present-biased respondents got it right (the difference is significant 1% level using a one-sided T-test, and at the 5% level using a two-sided T-test). This suggests to us that cognitive limitations may not play a significant role in driving the effect of present bias on the energy-efficiency gap.

Present-biased individuals may be choosing relatively inefficient motorcycles, either because they are undervaluing fuel economy and future cost savings, or because they may find it onerous to do the lifetime cost calculation, and computing lifetime costs requires exerting upfront effort. We already observed that present-biased individuals in our sample were less likely to be aware of fuel economy when they bought their motorcycles. Furthermore, they were more likely to state that they did not know what was to happen to future petrol prices than non-present-biased individuals. <sup>18</sup> In a similar vein, we find that about 65% of present-biased individuals stated that they did not know how long they planned to own their motorcycle (with respect to the purchase of the new motorcycle), as opposed to 58% of non-present-biased individuals (although the p-value of 0.13 with a one-sided test implies that the difference is only marginally significant). Moreover, we also find that about 32% of present-biased respondents stated that they were reimbursed for their petrol costs, as opposed to 22% of non-present-biased individuals (the difference being significant at the 5% level using a one-sided T-test and at the 10% level using a two-sided T-test). This correlational evidence further points to the fact that present-biased individuals may be further disincentivised from doing lifetime cost calculations. While we do not measure present bias in terms of effort in our study, this suggestive evidence points to the fact that present-biased (measured in the form of monetary rewards) individuals may also tend to be averse to computing lifetime costs. Therefore, we can speculate that our measure of present bias in terms of monetary payments approximates the level of present bias in terms of effort.

It may also be the case that present-biased individuals prioritize other aspects of the motorcycle more than its fuel economy. For example, we find that 64% of present-biased individuals, in a question in the survey on the importance of different attributes in forming their decision (for the new vehicle that they wanted to purchase), stated that the popularity of the motorcycle model among family and friends was very important to them, compared to 54% of non-present-biased individuals (and the difference is significant at the 5% level using a one-sided T-test, while it is insignificant using a two-sided T-test). It is possible that these individuals chose to overlook differences in fuel economy when they purchased their motorcycles.

On the other hand, we observe that none of the other behavioral anomalies are significant in determining the ratio of lifetime costs, across model specifications (in this sense, our findings are similar to Bradford et al. (2017), who also found a role for present bias in determining fuel-efficiency choices, but not for time preferences). Another finding that is consistent across

 $<sup>^{18}\</sup>mbox{About }4.3\%$  of present-biased individuals stated that they did not know how petrol prices would develop in the future, as opposed to 1.6% of non-present-biased respondents. This difference is significant at the 5% level using a one-sided T-test.

models in Table 3 is that income plays an important role in determining the ratio of total lifetime costs, with individuals having higher levels of income being associated with relatively higher total lifetime costs, compared to individuals who have low levels of income. Thus, richer individuals are likely to purchase more fuel-inefficient motorcycles compared to individuals who have lower levels of income. We observe this, having controlled for engine size as well as other motorcycle-related specifications in our model (such as brand, and age of the motorcycle). In addition, we find that respondents who stated that other members of their family also used the motorcycle on a regular basis were more likely to make relatively inefficient choices, and choose motorcycles that have higher total lifetime costs than the most efficient one, given the engine size. This result is also confirmed across model specifications.

Membership in clubs or groups is likely to result in relatively efficient motorcycle choices, as we find from the results of the IV-based estimations of columns (3), (4), (7) and (8). The negative coefficient suggests that some form of social ties and contacts may be associated with individuals preferring to adopt more efficient motorcycles. Interestingly, we also find across estimations that owning more than one motorcycle is correlated with individuals making relatively less efficient choices on the motorcycle that we asked them about in the survey (the motorcycle that they use most frequently), and this variable is significant at the 1% level across models.

Lastly, we also identify individuals who stated that they applied for a loan in order to purchase their motorcycles as having made relatively inefficient choices. We do not have information on whether they were actually granted a loan to buy the motorcycle, and while it is difficult to identify the underlying reason for this correlation without further information, this finding suggests that liquidity and credit may play a role in determining the efficiency of motorcycle choices as well (perhaps because these individuals, much like present-biased individuals, are likely to have focused on other attributes, and not on the fuel economy). In a question in our survey on the importance of different attributes in forming their decision f the new vehicle that they wanted to purchase, about 62% of respondents who had applied for a loan for their existing motorcycle stated that fuel economy was very important to them; this proportion was 84% for those who didn't apply for a loan (and this difference is significant at the 1% level using either one-sided or two-sided T-tests). By not paying attention to fuel economy, individuals who applied for loans may have ended up buying motorcycles having higher total lifetime costs.

However, it may also be the case that these individuals found it difficult to do the calculation of total lifetime costs, or that they didn't really care to do it; for example, we find that among those who applied for a loan, about 40% of respondents were able to use information on fuel economy to compute total fuel costs (in a hypothetical computational question), whereas 58% of people who didn't apply for a loan were able to solve this question (this difference is significant at the 1% level, using either a one-side or a two-sided T-test).

Thus, while we are able to find some suggestive support for our main results, the identification of exact channels of influence requires more detailed data.

The selection equation and first-stage estimation results corresponding to the Heckman and IV models in columns (2)-(4) and (6)-(8) of Table 3 are presented in Table 4. Column (1) presents the Heckman selection equation results corresponding to the models in columns (2) and (6) of Table 3, whereas columns (2) and (3) include the probit estimation results for the

IV-control function approach using one instrument (which corresponds to the first-stage of the models in columns (3) and (7) of Table 3), and the probit results for the control function estimation using two instruments (relevant for columns (4) and (8) of Table 3) respectively.

In column (1) of Table 4, we find that individuals who have children are more likely to own a motorcycle than those who did not; moreover, rather intuitively, income is another important determinant of the decision of households to purchase a motorcycle, with higher levels of income associated with higher likelihoods of owning a motorcycle. We find that both students as well as female respondents are less likely to already own a motorcycle, compared to non-students or male respondents. In this model, we use two external instruments to account for selection; the first is the number of bus stops within 500 metres of the centre of the locality where the respondent resides. The second is the number of motorcycle shops and showrooms within 500 metres of the locality centre. We find that these variable coefficients have a positive sign in this specification, even though they are insignificant.

From the probit results in column (2), we learn that individuals who stated that they had been affected by natural disasters were more likely to be present-biased than those who had not been affected, as expected. This variable serves as our instrumental variable. In addition, we find that individuals having higher discount factors (or lower discount rates) are more likely to be present-biased. This finding has been observed in previous studies; for instance, Schleich et al. (2019) find in their data from a sample of European countries that the average correlation coefficient between the annual discount factor and the 'beta' parameter that determines present bias (with lower values of beta representing present bias) is negative. Moreover, we find that individuals who are risk-averse or loss averse are also more likely to be present-biased. On the other hand, individuals who were aware of the fuel economy of the motorcycle when they purchased it were less likely to be present-biased. Having children is also a trait that is positively associated with being present-biased, among individuals who own a motorcycle, whereas higher levels of income are negatively associated with present bias. Lastly, we find that individuals who applied for a loan to purchase the vehicle were less likely to be present-biased; while we do not know whether these individuals were granted a loan or not, this result hints that liquidity constraints as a factor for present bias may be less relevant for our sample. The results of column (3) are similar, except that we introduce another instrumental variable, namely the intensity of the 2015 earthquake in the locality where the respondent lives. We find that while this variable has a positive coefficient (suggesting that individuals who lived in localities that were worse affected by the earthquake were more likely to be present-biased), it is insignificant in the model.

#### 4.3 Robustness checks

Table 7 in the appendix includes the results of robustness checks. In this table, we only report the results that test for the robustness of the model specifications in columns (4) and (8) of Table 3, i.e. the models estimated using the IV control function approach and two instruments. We choose these to be our baseline model results, as we address endogeneity concerns in these specifications, and we use both a subjective as well as an objective measure of the impact of natural disasters as instrumental variables. The results reported in this table are all also confirmed if we use the model specification with one instrumental variable (as in columns (3) and (7) of Table 3), these results can be provided on request.

**Table 4:** First-stage estimation results

Dependent variable Model	Probability of being selected Heckman (MLE)	Present bias indicator Probit (control function with 1 IV)	Present bias indicator Probit (control function with 2 IVs)
Column	(1)	(2)	(3)
Present bias indicator	-0.132		
	(0.124)		
Annual discount factor	0.248	1.412**	1.444**
Whether risk averse	(0.190) 0.086	(0.704) 0.621***	(0.692) 0.646***
Whiether fisk averse	(0.080)	(0.197)	(0.198)
Whether loss averse	-0.007	0.597**	0.601**
	(0.129)	(0.285)	(0.278)
Age of respondent	0.006	0.005	0.006
	(0.005)	(0.010)	(0.010)
Whether have children	0.146*	0.341*	0.324
	(0.085)	(0.209)	(0.211)
Monthly household income			
Between Rs. 30,000 and 50,000	0.260**	-0.644***	-0.669***
	(0.116)	(0.249)	(0.256)
Between Rs. 50,000 and 75,000	0.774***	-0.530***	-0.540***
M .I D 75.000	(0.101)	(0.231)	(0.233)
More than Rs. 75,000	0.620*** (0.127)	-1.012*** (0.276)	-1.046*** (0.279)
Whether student	-0.260***	-0.119	-0.096
vincinci stadent	(0.080)	(0.285)	(0.289)
Whether female	-0.663***	-0.522**	-0.506**
	(0.079)	(0.253)	(0.256)
Whether have a university degree	0.008	0.088	0.088
	(0.067)	(0.196)	(0.197)
Whether receives remittances	-0.116	0.231	0.227
Whether member of a club or society	(0.079) 0.055	(0.187) 0.152	(0.189) 0.156
Whether member of a club of society	(0.102)	(0.178)	(0.178)
Whether self/family member known to have a respiratory disorder	0.046	0.162	0.158
, . ,	(0.093)	(0.281)	(0.283)
Whether aware of fuel economy when bought motorcycle		-0.322**	-0.310**
		(0.142)	(0.147)
Whether correctly solved math question on lifetime costs		0.262**	0.260**
M/Lashan with home of Community Lands		(0.140)	(0.138)
Whether reimbursed for petrol costs		0.342 (0.213)	0.325 (0.217)
Whether family uses motorcycle regularly		-0.280*	-0.287*
Trictics raining asses motorcycle regularly		(0.172)	(0.170)
Whether applied for loan to purchase motorcycle		-0.730***	-0.727***
		(0.278)	(0.276)
Whether motorcycle is second-hand		0.244	0.225
M/L add an area and all area and area and a		(0.185)	(0.181)
Whether own more than one motorcycle		0.060 (0.358)	0.086 (0.354)
Whether self/someone known owned a vehicle during blockade	0.158**	(U.358) -0.444***	(0.354) -0.460***
	(0.067)	(0.174)	(0.175)
Number of bus stops within 500 metres	0.009	(5.2)	()
·	(0.034)		
Number of motorcycle shops within 500 metres	0.022		
Miles of the second sec	(0.020)	0.741***	0.702***
Whether affected by natural disasters and experienced severe damage		0.741***	0.723***
Intensity of 2015 earthquake (MMI)		(0.163)	(0.169) 0.464
mediaty of 2013 earthquake (MIMI)			(0.376)
Observations	2244	570	570

Note: The table reports the coefficients as well as standard errors in parentheses for the main explanatory variables. The sample in column (1) includes both current motorcycle owners as well as respondents who don't currently own a motorcycle, and comprises respondents who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles. In columns (2) and (3), it only includes current motorcycle owners. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample. All models include controls for districts and distance from the city centre, whereas the models in columns (2) and (3) also include controls for brand, engine size and age of the motorcycle. Standard errors are clustered at the locality level. \*,\*\* and \*\*\* respectively denote significance at 10%, 5% and 1% levels. Coefficient on constant is not reported.

In columns (1) to (3), we check the sensitivity of our main results to the choice of discount rate. In column (1), we present the second-stage estimation results of the IV-control function estimation in which the dependent variable is calculated assuming a discount rate of 3%, whereas in columns (2) and (3), it is calculated assuming rates of 5% and 9% respectively. We find that our main result on the effect of present bias on the energy-efficiency gap holds across models, on using varying discount rates.

In columns (4) to (7), we present the second-stage estimation results using alternative definitions of the risk aversion variable. As described in the Data section, for our main specifications, we converted the categorical variable capturing risk aversion and taking values from 0 to 10 into a dummy variable based on the median cutoff of 5. In columns (4) and (6), we use the  $25^{th}$  percentile as a cut-off to create the dummy variable, whereas in columns (5) and (7), we use the  $75^{th}$  percentile. Columns (4) and (5) are estimated using the ratio of lifetime costs computed using a discount rate of 7% as the dependent variable, whereas the models in columns (6) and (7) are estimated using a 10% discount rate. Across model specifications, we find that our main result on the positive effect of present bias on the ratio of total lifetime costs is confirmed. Moreover, we find that in the results of columns (5) and (7), the risk aversion dummy is significant at the 10% level, with a negative sign, suggesting that individuals who are risk averse are less likely to make inefficient motorcycle choices.

Lastly, in columns (8) and (9), we present the results estimated using a life time of 5 years to compute the total lifetime costs, instead of 10 years, as was used in the main specifications of Table 3, with a discount rate of 7% being used to calculate the lifetime cost ratio in column (8), and 10% used in column (9). Our main result on the positive effect of present bias is again valid for both model specifications, suggesting that varying the hypothetical duration of ownership of the motorcycle does not significantly hamper the magnitude of the effect of present bias on the energy-efficiency gap.

#### 5 Conclusion

Our study aims to shed light on the role of present bias, loss aversion, risk aversion as well as time preferences in determining the energy-efficiency gap associated with the choice of a durable in a low-income context, as well as on what factors determine the existence of these biases. Present bias has been found to be an important roadblock towards individuals in low-income countries undertaking investments in privately beneficial technologies and behaviors (such as in the adoption of preventative health measures). In this paper, we sought to investigate whether present bias may also impede purchases of fuel-efficient motorcycles in Kathmandu, Nepal.

We find that income is a strong determinant of time preferences, as well as risk and loss aversion in our sample. In addition, education, health status, as well as membership in groups or clubs also determined the occurrence of different anomalies. We find that certain psychological factors played a salient role in determining the existence these anomalies in Kathmandu. In subsequent analyses, we find that present-biased individuals are less likely to have bought more fuel-efficient motorcycles, and more likely to have purchased motorcycles having relatively higher total lifetime costs. This result is robust across several specifications. Moreover, we do not find a role of other biases (such as loss or risk aversion), but we do find that individuals who applied for loans to purchase a motorcycle were more likely to buy relatively inefficient

models, as were individuals who had higher levels of income or those who owned more than one motorcycle.

While our data does not allow us to sufficiently disentangle the channels for these observed effects, we hypothesize, and provide suggestive evidence, that present-biased individuals are likely to undervalue future energy cost savings, and find it burdensome to compute total lifetime costs given the substantial effort that it often warrants. Present bias can manifest itself as a preference for immediate rewards over later ones, but it can also imply that individuals find it onerous to undertake tasks in the present (such as computing total lifetime costs). In our study, we measure present bias in terms of monetary rewards, however we highlight some reasons as to why present-biased individuals may be more reluctant to calculate total lifetime costs (and this may result in purchasing relatively inefficient motorcycles). A more detailed exploration of the role of present bias in effort in hindering lifetime cost calculations (and thus possibly contributing to the energy-efficiency gap) makes a fruitful topic for future research.

Our results have important policy implications, given that standards as well as product taxes may be more relevant than fuel taxes in ensuring that present-biased individuals purchase more fuel-efficient vehicles. Moreover, liquidity constraints, that are also closely linked to present bias in this context, may need to be addressed as well in order to enable individuals to invest in fuel-efficient transport. This is particularly important in low and middle-income settings such as Nepal, where strong negative externalities from air pollution are a cause for concern, and other policy instruments such as standards and labels are yet to be implemented.

#### References

- Alberini, A., Di Cosmo, V. and Bigano, A. (2019), 'How are fuel efficient cars priced? Evidence from eight EU countries', *Energy Policy* **134**, 110978.
- Allcott, H. and Taubinsky, D. (2015), 'Evaluating Behaviorally Motivated Policy: Experimental Evidence from the Lightbulb Market', *American Economic Review* **105**(8), 2501–2538.
- Allcott, H. and Wozny, N. (2014), 'Gasoline Prices, Fuel Economy, and the Energy Paradox', *Review of Economics and Statistics* **96**(5), 779–795.
- Anderson, S. T. and Newell, R. G. (2004), 'Information programs for technology adoption: The case of energy-efficiency audits', *Resource and Energy Economics* **26**(1), 27–50.
- Angeletos, G. M., Laibson, D., Repetto, A., Tobacman, J. and Weinberg, S. (2001), 'The hyperbolic consumption model: Calibration, simulation, and empirical evaluation', *Journal of Economic Perspectives* **15**(3), 47–68.
- Augenblick, N., Niederle, M. and Sprenger, C. (2015), 'Working over Time: Dynamic Inconsistency in Real Effort Tasks', *The Quarterly Journal of Economics* **130**(3), 1067–1115.
- Bajracharya, I. and Bhattarai, N. (2016), 'Road Transportation Energy Demand and Environmental Emission: A Case of Kathmandu Valley', *Hydro Nepal: Journal of Water, Energy and Environment* **18**, 30–40.
  - **URL:** https://www.nepjol.info/index.php/HN/article/view/14641
- Berkouwer, S. B. and Dean, J. T. (2019), 'Credit, attention, and externalities in the adoption of energy efficient technologies by low-income households', *International Growth Centre Working Paper* (E- 47415-KEN-2).
- Booij, A. S. and van de Kuilen, G. (2009), 'A parameter-free analysis of the utility of money for the general population under prospect theory', *Journal of Economic Psychology* **30**(4), 651–666.
- Bradford, D., Courtemanche, C., Heutel, G., McAlvanah, P. and Ruhm, C. (2017), 'Time preferences and consumer behavior', *Journal of Risk and Uncertainty* **55**(2-3), 119–145. **URL:** https://doi.org/10.1007/s11166-018-9272-8
- Busse, M. R., Knittel, C. R. and Zettelmeyer, F. (2013), 'Are consumers myopic? Evidence from new and used car purchases', *American Economic Review* **103**(1), 220–256. **URL:** http://dx.doi.org/10.1257/aer.103.1.220
- Cameron, L. and Shah, M. (2015), 'Risk-taking behavior in the wake of natural disasters', Journal of Human Resources **50**(2), 484–515.

  URL: http://jhr.uwpress.org/content/50/2/484
- Can, B. and Erdem, O. (2013), 'Present-Bias in Different Income Groups', *Maastricht University* **RM/13/008**.
- Carvalho, L. S., Meier, S. and Wang, S. W. (2016), 'Poverty and Economic Decision-Making: Evidence from Changes in Financial Resources at Payday', *American Economic Review* **106**(2), 260–284.

- Cassar, A., Healy, A. and von Kessler, C. (2017), 'Trust, Risk, and Time Preferences After a Natural Disaster: Experimental Evidence from Thailand', World Development **94**, 90–105.
- Cassidy, R. (2018), 'Are the Poor so Present-Biased?', *IFS Working Paper* **18/24**, 1–41. **URL:** https://rachelmcassidy.com/research/.
- Cohen, F., Glachant, M. and Söderberg, M. (2017), 'Consumer myopia, imperfect competition and the energy efficiency gap: Evidence from the UK refrigerator market', *European Economic Review* **93**, 1–23.
- Dean, M. and Sautmann, A. (2021), 'Credit Constraints and the Measurement of Time Preferences', *The Review of Economics and Statistics* **103**(1), 119–135.
- Ding, X., Hartog, J. and Sun, Y. (2010), 'Can we measure Individual Risk Attitudes in a Survey?', *Tinbergen Institute discussion paper* **2010-027/3**. **URL:** https://ideas.repec.org/p/iza/izadps/dp4807.html
- Dohmen, T., Falk, A., Huffman, D., Sunde, U., Schupp, J. and Wagner, G. G. (2011), 'Individual risk attitudes: Measurement, determinants, and behavioral consequences', *Journal of the European Economic Association* **9**(3), 522–550.

**URL:** https://doi.org/10.1111/j.1542-4774.2011.01015.x

- Energy Policy Institute at the University of Chicago (2019), 'The Air Quality Life Index'. **URL:** https://aqli.epic.uchicago.edu/the-index/
- Espey, M. and Nair, S. (2005), 'Automobile fuel economy: What is it worth?', *Contemporary Economic Policy* **23**(3), 317–323.

**URL:** http://doi.wiley.com/10.1093/cep/byi024

- Facts Research and Analytics (2018), Facts of Nepal 2018 Edition.
- Falk, A., Becker, A., Dohmen, T., Huffman, D. B. and Sunde, U. (2016), 'The Preference Survey Module: A Validated Instrument for Measuring Risk, Time, and Social Preferences', *IZA Institute of Labor Economics Working Paper* **No. 9674**.

**URL:** https://ssrn.com/abstract=2725035

- Frederick, S., Loewenstein, G. and O'Donoghue, T. (2002), 'Time discounting and time preference: A critical review', *Journal of Economic Literature* **40**(2), 351–401.
- Fuerst, F. and Singh, R. (2018), 'How present bias forestalls energy efficiency upgrades: A study of household appliance purchases in India', *Journal of Cleaner Production* **186**, 558–569.
- Gerarden, T., Newell, R. G. and Stavins, R. N. (2015), 'Deconstructing the energy efficiency gap: Conceptual frameworks and evidence', **105**(5), 183–186. **URL:** http://dx.doi.org/10.1257/aer.p20151012
- Gillingham, K. and Palmer, K. (2014), 'Bridging the Energy Efficiency Gap: Policy Insights from Economic Theory and Empirical Evidence', *Review of Environmental Economics and Policy* **8**(1), 18–38.

- Golove, W. H. and Eto, J. H. (1996), 'Market Barriers to Energy Efficiency: A Critical Reappraisal of the Rationale for Public Policies to Promote Energy Efficiency', *Lawrence Berkeley National Laboratory*.
  - **URL:** https://emp.lbl.gov/publications/market-barriers-energy-efficiency
- Greene, D. L., Evans, D. H. and Hiestand, J. (2013), 'Survey evidence on the willingness of U.S. consumers to pay for automotive fuel economy', *Energy Policy* **61**, 1539–1550.
- Hardeweg, B., Menkhoff, L. and Waibel, H. (2013), 'Experimentally validated survey evidence on individual risk attitudes in rural Thailand', *Economic Development and Cultural Change* **61**(4), 859–888.
- Harding, M. and Hsiaw, A. (2014), 'Goal setting and energy conservation', *Journal of Economic Behavior and Organization* **107**(Part A), 209–227.
- Hausman, J. A. (1979), 'Individual Discount Rates and the Purchase and Utilization of Energy-Using Durables', *The Bell Journal of Economics* **10**(1), 33.
- Heckman, J. J. (1976), 'The Common Structure of Statistical Models of Truncation, Sample Selection and Limited Dependent Variables and a Simple Estimator for Such Models', *Annals of Economic and Social Measurement* **5**(4), 475–492.
- Heutel, G. (2019), 'Prospect theory and energy efficiency', *Journal of Environmental Economics* and Management **96**, 236–254.
- Hjorth, K. and Fosgerau, M. (2011), 'Loss Aversion and Individual Characteristics', *Environmental and Resource Economics* **49**(4), 573–596.
  - **URL:** https://link.springer.com/article/10.1007/s10640-010-9455-5
- Hunter, R. F., Tang, J., Hutchinson, G., Chilton, S., Holmes, D. and Kee, F. (2018), 'Association between time preference, present-bias and physical activity: Implications for designing behavior change interventions', *BMC Public Health* **18**(1), 1388.
  - **URL:** https://doi.org/10.1186/s12889-018-6305-9
- Jaffe, A. B. and Stavins, R. N. (1994), 'The Energy-efficiency Gap: What Does it Mean?', Energy Policy 22(10), 804–810.
- Kaur, S., Kremer, M. and Mullainathan, S. (2015), 'Self-Control at Work', *Journal of Political Economy* **123**(6), 1227–1277.
- Kremer, M., Rao, G. and Schilbach, F. (2019), Behavioral development economics, *in* 'Handbook of Behavioral Economics Volume 2', Elsevier, pp. 345–458.
  - **URL:** https://doi.org/10.1016/bs.hesbe.2018.12.002
- Laibson, D. (2011), 'Golden eggs and hyperbolic discounting', *The Quarterly Journal of Economics* **112**(2), 443–477.
- Lockwood, B. B. (2020), 'Optimal Income Taxation with Present Bias', *American Economic Journal: Economic Policy* **12**(4), 298–327.
- Meier, S. and Sprenger, C. (2010), 'Present-biased preferences and credit card borrowing', *American Economic Journal: Applied Economics* **2**(1), 193–210.

- Mullainathan, S. and Thaler, R. H. (2000), 'Behavioral Economics', *NBER Working Paper No.* 7948.
- Nayum, A. and Klöckner, C. A. (2014), 'A comprehensive socio-psychological approach to car type choice', *Journal of Environmental Psychology* **40**, 401–411.
- Nepal Drives (2020), 'Everest Bank Announces Bike Loan at 9.98% Per Annum'. **URL:** https://nepaldrives.com/everest-bank-bike-loan/
- Peters, A., Gutscher, H. and Scholz, R. W. (2011), 'Psychological determinants of fuel consumption of purchased new cars', *Transportation Research Part F: Traffic Psychology and Behaviour* **14**(3), 229–239.
- Sakha, S. (2019), 'Determinants of risk aversion over time: Experimental evidence from rural Thailand', *Journal of Behavioral and Experimental Economics* **80**, 184–198.
- Sallee, J. (2014), 'Rational Inattention and Energy Efficiency', *The Journal of Law and Economics* **57**(3), 781–820.
- Sallee, J. M., West, S. E. and Fan, W. (2016), 'Do consumers recognize the value of fuel economy? Evidence from used car prices and gasoline price fluctuations', *Journal of Public Economics* **135**, 61–73.
- Sawada, Y., Kuroishi, Y., Sawad, Y. and Kuroishi, Y. (2015), 'How does a natural disaster affect people's preference? The case of a large scale flood in the Philippines using the convex time budget experiments', *Disaster risks, social preferences, and policy effects: Field experiments in selected ASEAN and East Asian Coutries* pp. 57–84.
- Schleich, J., Gassmann, X., Meissner, T. and Faure, C. (2019), 'A large-scale test of the effects of time discounting, risk aversion, loss aversion, and present bias on household adoption of energy-efficient technologies', *Energy Economics* **80**, 377–393.
- Schwela, D. (2009), 'A Strategic Approach for Air pollution Reduction in Kathmandu Valley'. **URL:** https://mediamanager.sei.org/documents/Publications/Atmospheric/strategic-approach-airpollution-kathmandu air.pdf
- Shogren, J. F. and Taylor, L. O. (2008), 'On Behavioral-Environmental Economics', *Review of Environmental Economics and Policy* **2**(1), 26–44.
- Shrestha, S. R., Kim Oanh, N. T., Xu, Q., Rupakheti, M. and Lawrence, M. G. (2013), 'Analysis of the vehicle fleet in the kathmandu valley for estimation of environment and climate co-benefits of technology intrusions', *Atmospheric Environment* **81**, 579–590.
- Shupp, R., Loveridge, S., Skidmore, M., Lim, J. and Rogers, C. (2017), 'Risk, Loss, and Ambiguity Aversion after a Natural Disaster', *Economics of Disasters and Climate Change* 1(2), 121–142.
  - **URL:** https://link.springer.com/article/10.1007/s41885-017-0013-2
- Strotz, R. H. (1955), 'Myopia and inconsistency in dynamic utility maximization', *Review of Economic Studies* **23**(3), 165–180.
  - **URL:** https://academic.oup.com/restud/article-lookup/doi/10.2307/2295722

- Trading Economics (2020), 'Nepal Interest Rate: 2003-2020 Data'.
  - **URL:** https://tradingeconomics.com/nepal/interest-rate
- Train, K. (1985), 'Discount Rates in Consumers' Energy-related Decisions: A Review of the Literature', *Energy* **10**(12), 1243–1253.
- Turrentine, T. S. and Kurani, K. S. (2007), 'Car buyers and fuel economy?', *Energy Policy* **35**(2), 1213–1223.
- Ubfal, D. (2016), 'How general are time preferences? Eliciting good-specific discount rates', *Journal of Development Economics* **118**, 150–170.
- Wendling, Z. A., Emerson, J. W., Esty, D. C., Levy, M. A. and de Sherbinin, A. (2018), '2018 Environmental Performance Index'.
  - **URL:** https://epi.envirocenter.yale.edu/epi-topline
- Wooldridge, J. M. (2010), Econometric Analysis of Cross Section and Panel Data, 2nd edition., Cambridge: MIT Press.
- World Health Organization (2016), 'Air pollution levels rising in many of the world's poorest cities'.
  - **URL:** https://www.who.int/news/item/12-05-2016-air-pollution-levels-rising-in-many-of-the-world-s-poorest-cities
- Yesuf, M. and Bluffstone, R. A. (2009), 'Poverty, risk aversion, and path dependence in low-income countries: Experimental evidence from Ethiopia', *American Journal of Agricultural Economics* **91**(4), 1022–1037.

## **Appendix**

- 9. a) Suppose you have to make a hypothetical choice. Which of the following would you prefer?
  - A) Receive Rs 1000 in 6 months
  - B) Receive an amount between Rs 550 and Rs 980 in one week.
  - b) If 9B is selected,

You said you would prefer to receive an amount between Rs 550 and Rs 980 in one week. Please choose one of the options below that you would further prefer.

- A) Rs. 980 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- B) Rs. 940 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- C) Rs. 900 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- D) Rs. 860 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- E) Rs. 800 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- F) Rs. 700 in one week. Below this amount I would rather receive Rs 1000 in 6 months.
- G) Rs. 550 in one week. Below this amount I would rather receive Rs 1000 in 6 months.

Figure 1: Present Bias and Time Preferences: First MPL-based Question

- 10. a) Suppose you have to make another hypothetical choice. Which of the following would you prefer?
  - A) Receive an amount between Rs 550 and Rs 980 in 6 months and one week
  - B) Receive Rs 1000 in 12 months.
  - b) If 10A is selected,

You said you would prefer to receive an amount between Rs 550 and Rs 980 in 6 months and one week. Please choose one of the options below that you would further prefer.

- A) Rs. 980 in 6 months and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- B) Rs. 940 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- C) Rs. 900 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- D) Rs. 860 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- E) Rs. 800 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- F) Rs. 700 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.
- G) Rs. 550 in one week and one week. Below this amount I would rather receive Rs 1000 in 12 months.

Figure 2: Present Bias and Time Preferences: Second MPL-based Question

## **Calculating the Time Preference Parameters**

We calculated two preference parameters individually for each respondent using the questions presented in Figures 1 and 2, the discount factor as well as the present bias 'beta' parameter which is then used to construct the indicator variable.

In order to compute these values, we identified the "switch-points" for each respondent, which is the point at which he/she stated that they would prefer option B to option A, in each of the MPLs. Following the methodology adopted in Schleich et al. (2019), we assume that individuals are indifferent at the mean values of the lines between which they switched: for instance, if the respondent chose option B) first and then again option B) in response to the question presented in Figure 1 above, he or she is assumed to be indifferent between receiving Rs. 1'000 in 6 months and Rs. 940 in one week. Respondents who never switched, i.e. always chose A, were assumed to be indifferent between receiving option A and the first option in the second price list, and vice-versa for those respondents who immediately switched.

These two switch points were then used to compute the 6-month discount factor  $\delta$ , as well as the present bias  $\beta$  parameter, using the following equations:

$$u^*(x_{A1.1}) = \delta u^*(x_{B1.1}) \tag{5}$$

$$u^*(x_{A1.2}) = \delta \beta u^*(x_{B1.2}) \tag{6}$$

where A1.1 and B1.1 are the monetary amounts to which the respondent is indifferent in the first question, and A1.2 and B1.2 are the monetary amounts to which the respondent is indifferent in the second question.

In computing these parameters, we assume that respondents have monotonous preferences (and thus at most one switch-point in both the MPLs), and we assume that the utility function is linear in the monetary amount, i.e. we assume a risk-neutral utility function. The value of  $\delta$  derived by solving equation 5 above is raised to the power of 52/23 in order to compute the annual discount factor.

The parameter beta is used to identify individuals who are present-biased. A  $\beta$  value equal to one denotes a person who is neither present-biased, nor future-biased,  $\beta>1$  denotes a future-biased individual, whereas  $\beta<1$  denotes a present-biased individual.

Table 5: Summary Statistics of Vehicle Specifications

F (:)	01	A f l (l /l)
Engine Size (in cc)	Observations	Average fuel economy (km/l)
100	25	70.32
110	132	53.83
125	181	65.60
150	139	59.40
160	49	56.08
180	10	47
190	4	35
200	30	35
220	4	39.5
250	4	43
350	13	43.15
Year of purchase of motorcycle	Observations	Average fuel economy (km/l)
Year of purchase of motorcycle 2008	Observations 5	Average fuel economy (km/l) 65.6
2008	5	65.6
2008 2009	5 8	65.6 60.125
2008 2009 2010	5 8 3	65.6 60.125 56
2008 2009 2010 2011	5 8 3 18	65.6 60.125 56 64.22
2008 2009 2010 2011 2012	5 8 3 18 26	65.6 60.125 56 64.22 61.23
2008 2009 2010 2011 2012 2013	5 8 3 18 26 53	65.6 60.125 56 64.22 61.23 58.70
2008 2009 2010 2011 2012 2013 2014	5 8 3 18 26 53 53	65.6 60.125 56 64.22 61.23 58.70 61.98
2008 2009 2010 2011 2012 2013 2014 2015	5 8 3 18 26 53 53	65.6 60.125 56 64.22 61.23 58.70 61.98 59.13
2008 2009 2010 2011 2012 2013 2014 2015 2016	5 8 3 18 26 53 53 72 116	65.6 60.125 56 64.22 61.23 58.70 61.98 59.13 58.04

*Note:* The table reports the average values of fuel economy (measured in km/l) by engine size and by year of purchase of two-wheeler. The relevant sample size is 591 observations. The sample comprises respondents for whom vehicle specification data was available, and who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample.

Table 6: Determinants of Behavioral Anomalies and Time Preferences: Marginal Effects

Dependent variable Model	Present biased Probit	Annual discount factor Fractional response	Loss averse Probit	Risk averse Probit
Socio-demo <sub>i</sub>	graphic variables			
Monthly household income				
Between Rs. 30,000 and 50,000	0.018	-0.013	-0.031***	0.163***
	(0.022)	(0.012)	(0.011)	(0.040)
Between Rs. 50,000 and 75,000	0.028	-0.006	-0.030*	0.169***
	(0.028)	(0.014)	(0.017)	(0.066)
More than Rs. 75,000	-0.019	-0.045***	-0.030**	0.231***
	(0.021)	(0.015)	(0.015)	(0.060)
Whether have a university degree (Bachelor's or Master's)	-0.002	0.027***	0.0006	-0.116***
	(0.018)	(800.0)	(0.012)	(0.023)
Whether have children	0.018	0.014	-0.010	-0.002
	(0.019)	(0.012)	(0.013)	(0.040)
Whether student	-0.017	0.028***	0.001	0.003
	(0.019)	(0.012)	(0.012)	(0.031)
Age	0.001	0.0003	0.001	-0.001
	(0.001)	(0.0006)	(0.001)	(0.002)
Whether female	-0.010	-0.008	0.005	0.019
	(0.014)	(800.0)	(0.012)	(0.030)
Whether receive remittances	0.031**	0.015*	-0.018	-0.060**
	(0.015)	(0.009)	(0.016)	(0.032)
Whether have a respiratory disorder (self or close family)	0.065***	-0.038***	0.005	-0.005
	(0.023)	(0.015)	(0.016)	(0.044)
Whether member of a club or society	0.055***	0.025***	-0.029**	-0.005
	(0.019)	(0.010)	(0.013)	(0.044)
Whether self/someone known owned a vehicle during blockade	-0.038***	-0.044***	-0.015	0.017
	(0.015)	(0.013)	(0.013)	(0.032)
Log of distance of locality from the city centre	-0.001	-0.001***	0.003	0.007**
	(0.001)	(0.0006)	(0.002)	(0.003)
Psycholog	gical variables			
Live according to religious principles	0.027	0.006	0.042***	0.216***
to renglade humerhies	(0.019)	(0.009)	(0.012)	(0.031)
Hold on to my family's traditions	0.031**	-0.027***	0.055***	0.061**
2, ranning 5 additions	(0.017)	(0.011)	(0.015)	(0.031)
Go out a lot (e.g. dinners, parties, other leisure activities etc.)	0.0003	0.018	-0.004	-0.134***
(g. annotes, parties, other results detivities etc.)	(0.019)	(0.012)	(0.015)	(0.042)
Enjoy my life to the full	0.024	-0.044***	0.026	0.142**
g-yy 23 6110 1011	(0.024)	(0.015)	(0.019)	(0.066)
Likely to buy the latest technological innovation	0.036**	0.006	0.039***	0.124***
zmery to buy the latest technological innovation	(0.018)	(0.010)	(0.013)	(0.038)
Observations	2245	2245	2245	2245

Note: The table reports the marginal effects represented in terms of change in the dependent variable, for a unit change in each explanatory variable, computed at the means of the explanatory variables. The sample includes both current motorcycle owners as well as respondents who don't currently own a motorcycle, and comprises respondents who were the main decision-makers regarding purchase of durables in the household, as well as the main users of their motorcycles. We exclude respondents owning motorcycles with engine size 500 cc and above from our data sample. All models include dummies for districts. Standard errors are clustered at the locality level. \*,\*\* and \*\*\* respectively denote significance at 10%, 5% and 1% levels. Coefficient on constant is not reported.

**Table 7:** Robustness checks

Dependent variable	Log of ratic	o of lifetime costs w	Log of ratio of lifetime costs with varying discount rates	Log of ratio of lifetime costs	s with 7% discount rate		ts with 10% discount rate	Log of ratio of lifetime costs with 7% discount rate Log of ratio of lifetime costs with 10% discount rate	Log of ratio of lifetime costs with 10% discount rate
Specification	3%	2%	%6	Risk aversion $25^{th}$ percentile R	aversion $25^{th}$ percentile Risk aversion $75^{th}$ percentile	Risk aversior	Risk aversion $75^{th}$ percentile	Life of motorcycle of 5 years	Life of motorcycle of 5 years
	Ξ	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)
Present bias indicator	0.065*	0.065*	*990'0	0.066**	0.067**	**990.0	**290.0	**890.0	0.068**
	(0.038)	(0.038)	(0.037)	(0.035)	(0.037)	(0.035)	(0.036)	(0.037)	(0.037)
Annual discount factor	(0.019	(0.025)	0.016	(0.013	(0.0076)	0.012	0.010	0.013	0.015
Whether risk averse	-0.005	-0.005	-0.006	0.010	-0.025*	0.010	-0.024*	-0.006	-0.006
	(0.00)	(0.009)	(0.009)	(0.011)	(0.014)	(0.011)	(0.014)	(6000)	(0.009)
Whether loss averse	-0.005	-0.005	-0.005	-0.007	-0.006	-0.007	-0.006	-0.005	-0.005
Age of respondent	-0.0005	-0.0005	-0.0005	-0.0005	-0.0004	-0.0005	-0.0004	-0.0006	-0.0006
-	(0.0008)	(0.0008)	(0.0008)	(0.0008)	(0.000)	(0.0008)	(0.000)	(0.0008)	(0.0008)
Whether have children	-0.009 (0.012)	-0.009	-0.009	-0.008 (0.012)	-0.010 (0.012)	-0.008 (0.012)	-0.010	-0.008 (0.012)	-0.008 (0.012)
Monthly household income									
Between Rs. 30,000 and 50,000	0.023***	0.023**	0.022**	0.021**	0.027***	0.020**	0.027***	0.020***	0.020**
000 01 0 000 01 0 0 000 01 0 0 0 0 0 0	(0.010)	(0.010)	(0.010)	(0.00)	(0.011)	(0.009)	(0.010)	(0.010)	(0.010)
Between Ks. 50,000 and 75,000	0.043***	0.042***	0.042***	0.040***	0.049***	0.040***	0.048***	0.041***	0.041***
More than Rs. 75,000	0.049***	0.048***	0.048***	0.047***	0.054***	0.047***	0.054**	0.047***	0.046***
	(0.014)	(0.013)	(0.014)	(0.013)	(0.015)	(0.013)	(0.014)	(0.013)	(0.013)
Whether student	-0.001	-0.001	-0.002	-0.003	-0.003	-0.003	-0.003	-0.004	-0.004
Whether female	-0.002	-0.002	-0.003	-0.003	-0.0001	-0.003	-0.0007	-0.004	-0.004
	(0.000)	(0.000)	(0.000)	(0.009)	(0.010)	(0.000)	(0.010)	(0.008)	(0.008)
Whether have a university degree	-0.004	-0.004	-0.004	-0.003	-0.003	-0.003	-0.003	-0.004	-0.004
-	(0.009)	(0.009)	(0.009)	(0.009)	(0.009)	(0.000)	(0:000)	(0.008)	(0.008)
Whether receives remittances	0.003	0.003	0.002	0.004	0.002	0.004	0.002	0.002	0.002
Whether member of a club or society	-0.014***	-0.014**	-0.014**	-0.014***	-0.014**	-0.014***	-0.014**	-0.013**	-0.013**
	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(900:0)	(0.006)	(0.006)
Whether self/family member known to have a respiratory disorder	-0.018	-0.017	-0.016	-0.015	-0.015	-0.014	-0.014	-0.014	-0.014
	(0.012)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)
Wifetier aware of tuel economy witer bought motorcycle	(0.010)	(0.010)	(0.010)	(0011)	(0.010)	(0.011)	(0.010)	(6000)	(0.009)
Whether correctly solved math question on lifetime costs	-0.009	-0.009	-0.009	-0.009	-0.007	-0.009	-0.007	800:0-	-0.008
Wilcohow rotas harrond for notice	(0.010)	(0.010)	(0.010)	(0.010)	(0.009)	(0.010)	(600.0)	(0.009)	(0.009)
Whether remodes to perior costs	(0.013)	0.000	-0.000	(0.013)	(0.013)	(0.013)	0.013	-0.004	(0.012)
Whether family uses motorcycle regularly	0.020***	0.020***	0.020***	0.019***	0.017***	0.019***	0.017***	0.020***	0.020***
	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)
whether applied for loan to purchase motorcycle	0.036	0.036	(0.014)	0.034	0.03/	0.034	0.038	0.03/	0.03/
Whether motorcycle is second-hand	0.014	0.013	0.012	0.013	0.015	0.012	0.015	0.010	0.010
	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)
Whether self/someone known owned a vehicle during blockade	0.007	0.007	0.008	0.006	0.010	200.0	0.010	0.008	0.008
Whather own more than one motorcials	(0.007)	0.007)	(0.006)	(0.006)	(0.007)	(0.006)	(0.007)	(0.006)	(0.006)
	(0.014)	(0.014)	(0.014)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)
Observations	559	559	559	559	559	559	559	559	559
Cragg Donald F-statistic	62.838	62.838	62.838	55.049	59.444	55.049	59.444	62.838	62.838

Note: The table reports the coefficients as well as standard errors (in parentheses) for the main explanatory variables in the second-stage estimation results of the control function estimation using two instrumental variables. The sample comprises respondents for whom which specification data was available i.e., who own a motorcycles with engine size 500 cc and above from our data sample. All models include controls for district, as well as for brand, distance to the city centre, engine size and age of the motorcycles. Standard errors are clustered at the locality level. "." and "" respectively denote significance at 10%, 5% and 1% levels. Coefficient on constant is not reported.

38

#### Working Papers of the Center of Economic Research at ETH Zurich

(PDF-files of the Working Papers can be downloaded at www.cer.ethz.ch/research/working-papers.html).

- 21/353 M. Filippini, N. Kumar, S. Srinivasan Behavioral Anomalies and Fuel Efficiency: Evidence from Motorcycles in Nepal
- 21/352 V. Angst, C. Colesanti Senni, M. Maibach, M. Peter, N. Reidt, R. van Nieuwkoop Economic impacts of decarbonizing the Swiss passenger transport sector
- 21/351 N. Reidt Climate Policies and Labor Markets in Developing Countries
- 21/350 V. Britz, H. Gersbach Pendular Voting
- 21/349 E. Grieg
  Public opinion and special interests in American environmental politics
- 21/348 N. Ritter, J. A. Bingler

  Do homo sapiens know their prices? Insights on dysfunctional price mechanisms from a large fi
  eld experiment
- 20/347 C. Daminato, M. Filippini, F. Haufler Personalized Digital Information and Tax-favoured Retirement Savings: Quasi-experimental Evidence from Administrative Data
- 20/346 V. Britz, H. Gersbach Open Rule Legislative Bargaining
- 20/345 A. Brausmann, E. Grieg Resource Discoveries and the Political Survival of Dictators
- 20/344 A. Jo
  The Elasticity of Substitution between Clean and Dirty Energy with Technological Bias
- 20/343 I. van den Bijgaart, D. Cerruti The effect of information on market activity; evidence from vehicle recalls
- 20/342 H. Gersbach, R. Wattenhofer A Minting Mold for the eFranc: A Policy Paper
- 20/341 L. Bretschger Getting the Costs of Environmental Protection Right

- 20/340 J. A. Bingler, C. Colesanti Senni Taming the Green Swan: How to improve climate-related fi nancial risk assessments
- 20/339 M. Arvaniti, T. Sjögren Temptation in Consumption and Optimal Redistributive Taxation
- 20/338 M. Filippini, S. Srinivasan Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam
- 20/337 F. Böser, C. Colesanti Senni Emission-based Interest Rates and the Transition to a Low-carbon Economy
- 20/336 L. Bretschger, E. Grieg, P. J.J. Welfens, T. Xiong Corona Fatality Development and the Environment: Empirical Evidence for OECD Countries
- 20/335 M. Arvaniti, W. Habla The Political Economy of Negotiating International Carbon Markets
- 20/334 N. Boogen, C. Daminato, M. Filippini, A. Obrist Can Information about Energy Costs Affect Consumers Choices? Evidence from a Field Experiment
- 20/333 M. Filippini, N. Kumar, S. Srinivasan Nudging the Adoption of Fuel-Efficient Vehicles: Evidence from a Stated Choice Experiment in Nepal
- 20/332 L. Bretschger, E. Grieg Exiting the fossil world: The effects of fuel taxation in the UK
- 20/331 H. Gersbach, E. Komarov Research Bubbles
- 20/330 E. V. Dioikitopoulos, C. Karydas Sustainability traps: patience and innovation
- 19/329 M. Arvaniti, C. K. Krishnamurthy, A. Crepin Time-consistent resource management with regime shifts
- 19/328 L. Bretschger, K. Pittel
  Twenty Key Questions in Environmental and Resource Economics
- 19/327 C. Karydas, A. Xepapadeas Climate change financial risks: pricing and portfolio allocation
- 19/326 M. Filippini, S. Srinivasan Investments in Worker Health and Labor Productivity: Evidence from Vietnam

- 19/325 H. Gersbach
  - Democratizing Tech Giants! A Roadmap
- 19/324 A. Brausmann, M. Flubacher and F. Lechthaler Valuing meteorological services in resource-constrained settings: Application to small-holder farmers in the Peruvian Altiplano
- 19/323 C. Devaux and J. Nicolai Designing an EU Ship Recycling Licence: A Roadmap
- 19/322 H. Gersbach Flexible Majority Rules for Cryptocurrency Issuance
- 19/321 K. Gillingham, S. Houde and A. van Benthem Consumer Myopia in Vehicle Purchases: Evidence from a Natural Experiment
- 19/320 L. Bretschger Malthus in the Light of Climate Change
- $19/319\,$  J. Ing and J. Nicolai Dirty versus Clean Firms' Relocation under International Trade and Imperfect Competition
- 19/318 J. Ing and J. Nicolai North-South diffusion of climate-mitigation technologies: The crowding-out effect on relocation
- 19/317 J. Abrell, M. Kosch and S. Rausch How Effective Was the UK Carbon Tax? - A Machine Learning Approach to Policy Evaluation
- 19/316 D. Cerruti, C. Daminato and M. Filippini The Impact of Policy Awareness: Evidence from Vehicle Choices Response to Fiscal Incentives
- 19/315 M. Filippini, N. Kumar and S. Srinivasan Energy-related financial literacy and bounded rationality in appliance replacement attitudes: Evidence from Nepal